



Robin Chataut ¹,*¹, Mary Nankya ² and Robert Akl ³

- ¹ School of Computing and Engineering, Quinnipiac University, Hamden, CT 06518, USA
- ² Department of Computer Science, Fitchburg State University, Fitchburg, MA 01420, USA; mnankya@fitchburgstate.edu
- ³ Department of Computer Science and Engineering, University of North Texas, Denton, TX 76203, USA; robert.akl@unt.edu
- * Correspondence: robin.chataut@qu.edu

Abstract: In the rapidly evolving landscape of wireless communication, each successive generation of networks has achieved significant technological leaps, profoundly transforming the way we connect and interact. From the analog simplicity of 1G to the digital prowess of 5G, the journey of mobile networks has been marked by constant innovation and escalating demands for faster, more reliable, and more efficient communication systems. As 5G becomes a global reality, laying the foundation for an interconnected world, the quest for even more advanced networks leads us to the threshold of the sixth-generation (6G) era. This paper presents a hierarchical exploration of 6G networks, poised at the forefront of the next revolution in wireless technology. This study delves into the technological advancements that underpin the need for 6G, examining its key features, benefits, and key enabling technologies. We dissect the intricacies of cutting-edge innovations like terahertz communication, ultra-massive MIMO, artificial intelligence (AI), machine learning (ML), quantum communication, and reconfigurable intelligent surfaces. Through a meticulous analysis, we evaluate the strengths, weaknesses, and state-of-the-art research in these areas, offering a wider view of the current progress and potential applications of 6G networks. Central to our discussion is the transformative role of AI in shaping the future of 6G networks. By integrating AI and ML, 6G networks are expected to offer unprecedented capabilities, from enhanced mobile broadband to groundbreaking applications in areas like smart cities and autonomous systems. This integration heralds a new era of intelligent, selfoptimizing networks that promise to redefine the parameters of connectivity and digital interaction. We also address critical challenges in the deployment of 6G, from technological hurdles to regulatory concerns, providing a holistic assessment of potential barriers. By highlighting the interplay between 6G and AI technologies, this study maps out the current landscape and lights the path forward in this rapidly evolving domain. This paper aims to be a cornerstone resource, providing essential insights, addressing unresolved research questions, and stimulating further investigation into the multifaceted realm of 6G networks. By highlighting the synergy between 6G and AI technologies, we aim to illuminate the path forward in this rapidly evolving field.

Keywords: 6G; 5G; artificial intelligence; machine learning; terahertz communication; ultra-massive MIMO; quantum communication; millimeter waves; blockchain; Internet of Things

1. Introduction

As globalization advances, the volume of mobile data traffic is experiencing a rapid and exponential increase. According to a report by the ITU-R, global mobile data traffic was 158 exabytes per month in 2022 and is projected to reach 2194 exabytes per month by 2028 and 5016 exabytes per month by 2030 [1]. These numbers represent an exponential increase in the amount of data consumed by mobile subscribers, with each subscriber projected to consume 257 gigabytes of data in 2030 compared to 12.1 gigabytes in 2022 [2].



Citation: Chataut, R.; Nankya, M.; Akl, R. 6G Networks and the AI Revolution—Exploring Technologies, Applications, and Emerging Challenges. *Sensors* **2024**, *24*, 1888. https://doi.org/10.3390/s24061888

Academic Editor: Peter Han Joo Chong

Received: 8 December 2023 Revised: 1 March 2024 Accepted: 11 March 2024 Published: 15 March 2024



Copyright: © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). The growing demand for mobile data services is not limited to a particular region or demographic [3]. By 2025, around 70% of the global population will utilize mobile services, with approximately 60% accessing mobile internet. This growth is further propelled by the proliferation of new technologies such as the Internet of Things, AI, blockchain, augmented and extended reality, 3D video, and connected vehicles [4].

5G technology has been deployed worldwide [5] to meet the increasing need for mobile data services. However, with the world moving towards automation, it is apparent that a more advanced technology than current 5G networks will be required to handle the rising data traffic [6]. This is where the sixth generation '6G' network comes in, which is expected to provide users with high-quality service while coping with this exponential increment in data traffic [7,8]. The sixth-generation network promises to be a game-changer in mobile wireless technology, with its ultra-fast data speeds, low latency, and massive connectivity. 6G networks will transform mobile networks by integrating AI and ML to seamlessly combine the physical, digital, and biological worlds. This integration will enable the creation of new use cases and applications that were not previously possible with 5G networks. Moreover, 6G networks will lay the foundation for developing smart cities, autonomous vehicles, and other applications that require reliable, high-bandwidth, and low-latency connectivity [9]. In this paper, we take a hierarchical approach to 6G networks and present a comprehensive overview of the 6G networks. Our summary of contributions and paper organization is as follows.

Section 2 delves into the technological foundations of 6G networks and the AI revolution, spanning from the inception of 1G to the cutting-edge developments of 6G and AI-revolutionized 6G networks and their anticipated impact on various sectors. In Section 3, we explore the distinctive features that define 6G networks, shedding light on the anticipated capabilities and innovations that set them apart from their predecessors. Section 4 discusses the integration of AI and ML in 6G networks. Section 5 addresses the pressing question of "When Will 6G Come Out?" by examining current timelines, ongoing research initiatives, and industry expectations surrounding the deployment of 6G technology. Section 6 discusses the convergence of AI, 6G, and wireless communication, highlighting enhanced mobile broadband (eMBB), ultra-reliable low-latency communication (URLLC), and massive machine-type communication (mMTC) use cases investigating the diverse applications of 6G networks, envisioning the transformative impact on industries, services, and everyday life. Section 7 scrutinizes the challenges associated with the deployment of 6G, ranging from technological hurdles to regulatory considerations, providing a comprehensive assessment of potential obstacles. In Section 8, we explore the key technologies that shape the deployment of the 6G network. Section 9 examines whether 6G poses health risks, delving into existing research surrounding the potential dangers of advanced wireless technologies and AI involvement in the mentioned key enabling technologies. Section 10 opens the door to future research endeavors by outlining unique recommendations and future research directions. Section 11 briefly touches upon the speculative realm of 7G networks, contemplating the potential directions and features that may define the next frontier in wireless communication. Finally, Section 12 concludes the comprehensive analysis, summarizing key findings and insights derived from the exploration of 6G networks and setting the stage for further research and development in the field. Figure 1 below shows the organizational structure for this paper.



Figure 1. Organizational structure of the paper.

2. Technological Foundations of 6G Networks and AI Revolution

The era of mobile communication started in the early 1980s and has seen significant development and expansion in the decades that followed. The advancement of mobile wireless technology can be divided into distinct eras, each of which has brought about substantial progress and developments in data rates, connectivity, and functionality [10]. The initial phase of mobile wireless technology, 1G, was introduced in the early 1980s and was primarily based on analog technology [11]. This generation of technology was primarily utilized for voice communication and was distinguished by its low data transfer speeds and subpar audio quality [12]. Some examples of 1G include Advanced Mobile Phone System (AMPS), Total Access Communication System (TACS), and Nordic Mobile Telephone (NMT) [13].

The introduction of second-generation (2G) mobile networks in the early 1990s marked a shift from analog to digital technology [14]. Along with traditional voice services, 2G networks introduced new capabilities such as short message service (SMS) and basic email functionality. 2G networks also improved audio quality and enhanced security [15]. Some of the well-known 2G (second generation) mobile networks include GSM (Global System for Mobile Communications), IS-95 (Interim Standard-95) [16] PDC (Personal Digital Cellular), and CDMAone (Code Division Multiple Access) [17].

Next was the introduction of 3G (third generation) mobile networks in the early 2000s, which marked a significant advancement in mobile technology, providing both voice and data services [18]. These networks offered elevated data transfer speeds and the capability of web browsing on mobile devices. They also introduced multimedia message support (MMS) and the ability to use data-intensive applications such as email, web browsing, video streaming, and mobile television [19]. In addition to providing enhanced data transfer speeds and web browsing capabilities, 3G networks expanded the coverage area and incorporated security measures such as packet data confidentiality and integrity. Some examples of 3G (third generation) mobile networks include CDMA2000 (Code Division Multiple Access 2000), WCDMA (Wideband Code Division Multiple Access), and EDGE (Enhanced Data rates for GSM Evolution) [20].

The 4G (fourth generation) mobile networks in the early 2010s marked a significant advancement in mobile technology, offering high data transfer speeds and improved network coverage [21]. These networks enabled HD video streaming, mobile video conferencing, online gaming, and high-speed mobile internet. Examples of 4G (fourth generation) mobile networks include LTE (Long-Term Evolution) and WiMAX (Worldwide Interoperability for Microwave Access) [22].

The introduction of 5G (fifth generation) mobile networks in the early 2010s represents the latest advancement in mobile technology, with the first 5G mobile towers coming online in 2018 [23]. These networks are distinguished by extremely high data transfer speeds, improved network coverage, and ultra-low latency. 5G networks are expected to be a foundation for the Internet of Things (IoT), smart cities, and the fourth industrial revolution [24].

While previous generations of wireless networks have already leveraged AI for optimization and automation, 6G takes this collaboration to unprecedented levels, integrating AI at every layer of the network architecture.

6G networks are currently being researched and developed as the next evolution of mobile networks, with the expectation of providing unparalleled transmission speeds, ultralow latency, and improved coverage [25]. These networks will incorporate cutting-edge technologies such as terahertz communication, ultra-massive MIMO, AI, machine learning (ML), quantum communication, millimeter, reconfigurable intelligent surfaces, etc. Potential applications for 6G networks include Linked robotic and self-governing systems, wireless brain–computer interfaces, blockchain advancements, immersive multi-sensory realities, space and deep-sea exploration, tactile internet capabilities, and industrial networking. Figure 2 shows the evolution of mobile communications.



Figure 2. Evolution of mobile communications: a chronological depiction of the advancements in mobile network technology from 1G in 1981 to the expected 6G in 2030. This visual encapsulates the major milestones in mobile communications, including the emergence of 2G and the introduction of SMS in 1992, the advent of 3G and mobile data in 2001, the expansion to 4G and high-speed internet access in 2011, and the integration of IoT with 5G in 2020. The future projection of 6G suggests a paradigm shift to smarter, AI-driven networks supporting 3D internet and enhanced video capabilities.

AI-Revolutionized 6G Networks and Their Anticipated Impact on Various Sectors

The convergence of 6G networks and AI technologies holds immense promise for transforming various sectors and unlocking new opportunities for innovation, efficiency, and economic growth. By harnessing the synergies between these two transformative technologies, we stand on the brink of a new era of connectivity, where the boundaries between the physical and digital worlds blur and the possibilities are limited only by our imagination. 6G is envisioned as a paradigm shift, introducing revolutionary capabilities that will reshape industries, economies, and societies at large, as seen below:

- Healthcare: 6G networks, coupled with AI, are poised to revolutionize healthcare delivery [26]. From remote patient monitoring to real-time telemedicine consultations, the ultra-low latency and high reliability of 6G will enable life-saving applications, such as remote surgery performed by robotic systems guided by AI algorithms. Furthermore, AI-driven predictive analytics can enhance disease detection and treatment outcomes, leading to more personalized and efficient healthcare services [27].
- **Transportation:** In the transportation sector, 6G networks will underpin the proliferation of autonomous vehicles and smart transportation systems [28]. AI-powered algorithms will leverage real-time data from vehicle-to-everything (V2X) communication to optimize traffic flow, enhance safety, and reduce congestion. Moreover, 6G's ultra-reliable low-latency communication (URLLC) capabilities will enable split-second decision making, ensuring the seamless operation of autonomous vehicles even in the most demanding scenarios.
- Manufacturing: With the rise of Industry 4.0, 6G networks will catalyze the transition towards smart factories characterized by interconnected, AI-driven production systems. By leveraging 6G's massive connectivity and AI's predictive maintenance capabilities, manufacturers can optimize production processes, minimize downtime, and maximize resource efficiency [29]. Collaborative robots equipped with AI algorithms will enable agile and flexible manufacturing workflows, adapting in real time to changing demand and market dynamics.
- Entertainment and media: The convergence of 6G networks and AI will revolutionize the entertainment and media landscape, ushering in immersive experiences and personalized content delivery. From augmented reality (AR) and virtual reality (VR) applications to AI-generated content recommendations, 6G networks will enable seamless, high-fidelity multimedia streaming with minimal latency [30]. AI algorithms will analyze user preferences and behavior in real time, delivering hyper-personalized content tailored to individual tastes and preferences.
- **Finance and banking:** In the financial sector, 6G networks will facilitate the proliferation of AI-driven fintech solutions, revolutionizing payment systems, fraud detection, and risk management. Real-time data analytics powered by AI will enable financial institutions to detect fraudulent transactions with unprecedented accuracy. At the same time, 6G's ultra-low latency will ensure near-instantaneous transaction processing, enhancing the efficiency and security of digital payments [31].

3. Features of 6G

6G networks are expected to bring significant advancements over the current 5G technology. The features and advantages of 6G networks are summarized in Figure 3. Here are some of the key features and potential advancements associated with 6G:

- High data transfer rates: 6G networks are expected to bring tremendous advancements in data transfer speeds, potentially reaching up to 10 Tbps. This represents a significant increase when compared to the current data transfer speed set for 5G networks, which is 10 Gbps [32].
- Low latency: 6G networks are expected to provide ultra-low latency, potentially reaching as low as 0.1 ms, which significantly improves upon the latency of 5G networks with a latency requirement of 1 ms [33].
- **Extended coverage:** 6G networks are expected to have an extended coverage range, potentially reaching deep-sea, space, and underground areas. This would enable the use of new applications such as deep-sea sightseeing, space travel, and industrial internet [34].
- Enhanced user experience: 6G networks are projected to enhance the user experience by amplifying the capabilities of extended reality, augmented reality, virtual reality, and AI [35].
- **Increased spectral efficiency:** 6G networks are expected to offer spectral and network efficiency ten times greater than that of 5G networks [36].

- Ubiquitous connection: 6G networks are expected to provide enormous broadcasting data and to support more than 1 million connections, which is a hundred times more than current 5G networks [37].
- **Better energy efficiency:** 6G networks are expected to have an optimized energy consumption, resulting in longer battery life, making it more sustainable and efficient to use [38].
- **Integration with other technology:** Anticipated integration of 6G networks involves seamless incorporation with other technologies, such as IoT, cloud computing, and big data analytics, ensuring efficient connections across various systems [39].

Figure 3 shows features of 6G network.



Figure 3. Features of 6G network.

Table 1 shows the performance comparison of 4G, 5G, and 6G networks [40].

Table 1. Comparison of 4G, 5G, and 6G performance indicators.

| Performance Indicator | 4 G | 5G | 6G |
|----------------------------------|---------------------|--------------------------------|--------------------------------|
| Maximum data transfer rate | 100 Mbps | 10 Gbps | Up to 10 Tbps |
| Minimum latency | 10 ms | 1 ms | Up to 0.1 ms |
| Maximum device density per sq km | 0.1 million devices | 1 million devices | 10 million devices |
| Energy efficiency | $1 \times$ | 100× more efficient than 4G | 100× more efficient than 5G |
| Spectral efficiency | 1× | 100× more efficient than 4G | 100× more efficient than 5G |
| Available spectrum | Up to 6 GHz | Up to 300 GHz | Up to 3 THz |
| Maximum mobility | 200 km/h | 300 km/h | 600 km/h |
| AI integration | None | Partial | Full |

4. AI and ML for 6G Networks

AI and ML are anticipated to have a revolutionary impact on 6G networks, improving network management optimization and enhancing the user experience across various aspects. AI is expected to play a crucial role in developing 6G networks by addressing the primary challenge of managing the significant increase in connected devices and data traffic [41]. By optimizing network resources like bandwidth and computing power, AI can guarantee the efficient operation of the network [42]. AI can enable real-time decisionmaking capabilities, essential for applications like autonomous vehicles and augmented reality to operate safely and efficiently. Furthermore, AI can create intelligent networks that can self-learn and adapt in real time by analyzing data to improve performance and efficiency. It can predict potential issues by analyzing sensor data and other sources, preventing service disruptions. This predictive maintenance for 6G networks reduces downtime and increases reliability [43]. On the other hand, ML algorithms can analyze user behavior patterns and preferences to personalize the network experience. This includes adaptive content delivery, predictive caching, and personalized service recommendations. AI-driven cybersecurity measures can continuously analyze network traffic patterns, detect anomalies, and proactively respond to security threats [44]. ML models can evolve and adapt to new cyber threats, enhancing overall network security. AI can also be applied to cognitive radio systems, allowing networks to autonomously adapt to changing radio conditions, interference, and spectrum availability [45]. This enables the more flexible and intelligent use of available radio resources [46]. Moreover, AI-driven techniques can enhance the resilience of 6G networks by predicting and mitigating the impact of faults or disruptions. ML models can adaptively reroute traffic and optimize network performance during failures [47].

ML will play a crucial role in enabling the development of 6G networks by providing intelligent and adaptive capabilities that can support various applications and services [48]. These algorithms can dynamically manage and optimize network resources in real time, depending on the network's specific use case and requirements. Reinforcement learning (RL) algorithms, such as Q-learning and Deep Q Networks (DQNs), can make sequential decisions in dynamic environments. In network management, RL can be applied to optimize resource allocation, routing, and scheduling based on changing conditions [49]. A multiobjective-based genetic algorithm is an optimization algorithm inspired by natural selection that can be used to solve resource allocation problems in networks, adapting to changing demands and constraints [50]. Particle swarm optimization (PSO) is a population-based optimization algorithm that models the social behavior of particles. It can be employed for dynamic resource allocation, load balancing, and network optimization [51]. Deep learning models like neural networks can undergo training to forecast network performance, recognize patterns, and enhance resource allocation [52]. Deep reinforcement learning (DRL) combines deep learning with RL for complex decision making. Fuzzy logic can be applied to model and control network parameters in a dynamic environment. It provides a way to handle uncertainty and imprecise information in network optimization [53]. Markov decision processes (MDPs) are used in RL to model decision-making problems with sequential interactions. In network optimization, MDPs can represent the dynamic nature of resource allocation and routing decisions [54]. Swarm intelligence algorithms inspired by swarm intelligence, such as the bee algorithm or the firefly algorithm, can be applied to optimize network resources collaboratively [55].

5. When Will 6G Come Out?

Anticipations suggest that 6G networks could debut around 2030, potentially emerging earlier in specific global regions. 6G wireless technology is currently the focus of research and development by several countries, universities, and tech companies worldwide. China has set an ambitious goal of dominating the 6G industry by 2030, and companies like Huawei, ZTE, and China Mobile are actively involved in 6G research [56–59]. In South Korea, LG has established a 6G research center [60], while Finland's University of Oulu is leading 6G research with the country's "6G Flagship" program [61]. The US Federal Communications Commission (FCC) has opened the "terahertz wave" for experiments on next-generation standards, which could include 6G. At the same time, companies like Qualcomm and Intel are also involved in 6G research [62]. Japan's 6G research program focuses on developing technology for super-fast data transfer rates, and the EU's Horizon 2020 5G-DRIVE project explores the potential of 6G [63,64]. However, despite the significant efforts being made, 6G technology is still in its early stages of development, and it may be some time before 6G networks become a reality.

6. The Convergence of AI, 6G, and Wireless Communication

With 6G technology, a wide array of possibilities unfolds across three key services: enhanced mobile broadband (eMBB), ultra-reliable low-latency communication (URLLC), and massive machine-type communication (mMTC) [65]. The above use cases are described in detail below:

- (a) Enhanced mobile broadband (eMBB): 6G networks are expected to further improve upon the enhanced mobile broadband capabilities of 5G by delivering even higher data rates, lower latency, and increased capacity [66]. Some potential aspects of eMBB in 6G networks include the following:
 - Ultra-high data rates: 6G networks could achieve significantly higher data rates compared to 5G, potentially reaching terabits per second (Tbps) speeds [67]. This would enable seamless streaming of immersive, high-resolution content such as holographic videos, uncompressed 8K or 16K video streaming, and ultra-HD virtual reality (VR) and augmented reality (AR) experiences.
 - Low-latency communication: 6G networks aim to further reduce latency to nearreal-time levels, enabling ultra-responsive applications such as cloud gaming, remote surgery, and autonomous vehicles. With latency reduced to microseconds or even nanoseconds, users can experience seamless interactions with remote systems and devices [68].
 - Massive capacity: 6G networks are expected to support a massive increase in connected devices and simultaneous connections, facilitating the proliferation of IoT devices, wearable technologies, and smart sensors [69]. This would enable seamless connectivity and data exchange in densely populated areas or scenarios with a high density of connected devices.
- (b) Ultra-reliable low-latency communication (URLLC): 6G networks will build upon the ultra-reliable low-latency communication capabilities of 5G by further reducing latency and increasing reliability. Key aspects of URLLC in 6G networks include the following:
 - Mission-critical applications: 6G networks will support mission-critical applications that require ultra-low latency and high reliability, such as industrial automation, remote surgery, and autonomous vehicles. By reducing latency to sub-millisecond levels and ensuring ultra-reliable communication links, 6G networks will enable seamless connectivity and real-time responsiveness in critical scenarios [70].
 - Predictive maintenance: With advanced analytics and AI integration, 6G networks can enable predictive maintenance in industrial settings, allowing machines and equipment to communicate in real time and anticipate maintenance needs before failures occur. This proactive approach to maintenance can minimize downtime, reduce operational costs, and optimize asset performance [71].
- (c) Massive machine-type communication (mMTC): 6G networks will continue to support massive machine-type communication, catering to the connectivity needs of many IoT devices and sensors. Key aspects of mMTC in 6G networks include the following:
 - Massive scalability: 6G networks will be designed to support a massive scale of connected devices, ranging from billions to trillions of IoT devices and sensors [72]. This scalability will enable the deployment of IoT solutions in various domains, including smart cities, industrial automation, agriculture, healthcare, and environmental monitoring.
 - Energy-efficient communication: 6G networks will incorporate energy-efficient communication protocols and techniques to optimize power consumption in IoT

devices and extend battery life [73]. This will enable long-lasting and sustainable IoT deployments, particularly in remote or inaccessible locations where power sources are limited.

 Diverse use cases: 6G networks will support diverse IoT applications with varying requirements in terms of data rates, latency, reliability, and energy consumption [74]. These applications include smart grids, asset tracking, environmental monitoring, smart agriculture, and smart healthcare, among others.

Table 2 compares 5G and AI-revolutionized 6G technology across enhanced mobile broadband, ultra-reliable low-latency communication, and massive machine-type communication.

Table 2. A comparison between 5G and AI revolutionized 6G technology across eMBB, URLLC, and mMTC.

| Technology | Enhanced Mobile Broadband (eMBB) | Ultra-Reliable Low-Latency Communication (URLLC) | Massive Machine-Type Communication (mMTC) |
|----------------------|---|---|---|
| 5G | Provides high data rates for mobile users, enabling high-definition video streaming [75]. | Offers reliable and low-latency communication, crucial for applications such as industrial automation and remote surgery [76]. | Supports a large number of connected devices, enabling efficient communication between a massive number of IoT devices [77]. |
| AI-revolutionized 6G | Leverages AI for better spectrum utilization and intelligent resource allocation [78]. | Improves URLLC with even lower latency and higher reliability through AI-powered network optimization and predictive maintenance capabilities [79]. | Optimizes resource allocation and communication protocols for efficient device-to-device communication and network slicing [80]. |

6.1. Applications of 6G Network

The 6G network is not yet commercially available; however, it is expected to have applications in several domains [81].

Figure 4 below shows the applications of 6G networks.



Figure 4. Applications of 6G networks.

6.1.1. Ultra-High-Resolution Video Streaming and Cloud Gaming

The advancement of the 6G network and the AI revolution will enable seamless streaming of ultra-high-resolution video content and cloud gaming experiences with unprecedented quality and minimal interruptions [82]. Complementing this infrastructure,

the AI revolution will leverage sophisticated algorithms to optimize streaming parameters in real time, ensuring smooth playback and personalized content recommendations based on user preferences. Together, this integration of 6G networks and AI technologies will revolutionize the entertainment landscape, offering users immersive, high-quality experiences that redefine the boundaries of mobile communication and AI.

6.1.2. Healthcare Monitoring

The convergence of AI and the 6G revolution is poised to revolutionize health monitoring applications by enabling real-time data processing, remote monitoring, personalized healthcare, predictive analytics, healthcare automation, and enhanced diagnosis and treatment [83]. The ultra-low latency and high data transfer speeds offered by 6G networks will facilitate the seamless transmission and analysis of health data, allowing for timely interventions and personalized healthcare recommendations based on individualized data insights. AI algorithms will be crucial in analyzing vast datasets to identify patterns and correlations, enabling predictive analytics for forecasting health trends and automating various healthcare processes [84]. This synergy between AI and 6G technologies promises to optimize patient care by providing more efficient, proactive, and personalized healthcare delivery.

6.1.3. Remote Surgery

The emergence of telesurgery, enabled by robotic technology and wireless networking, revolutionizes surgical practice by connecting geographically distant patients and surgeons [85]. This innovative approach harnesses the power of advanced telecommunications and robotics, facilitated by the anticipated 6G network revolution and artificial intelligence technologies. The 6G network's ultra-fast data transmission speeds and ultralow latency ensure seamless communication between remote robotic surgical systems and surgeons, overcoming geographical barriers and providing high-quality surgical care in remote locations [86]. AI-powered robotic systems exhibit precise and dexterous movements, augmented by AI algorithms that continuously learn and adapt to improve surgical outcomes. Real-time data processing and analysis by AI algorithms enable remote surgeons to make informed decisions during telesurgery procedures, enhancing patient safety and surgical precision. Overall, integrating 6G networking capabilities and AI technologies in telesurgery advances surgical practice by improving access to care, enhancing surgical outcomes, and ensuring patient safety.

6.1.4. Smart Grid

Due to the integration of 6G and artificial intelligence, a significant revolution is underway in the development and functionality of smart grids. 6G networks are anticipated to offer ultra-fast data transmission speeds, ultra-low latency, and extensive device connectivity, thereby significantly improving the communication infrastructure of smart grids [87]. This will enable the real-time monitoring, control, and optimization of energy distribution and consumption within the grid, facilitating efficient energy management and response to dynamic demand patterns. AI technologies, when integrated with 6G networks, further enhance the capabilities of smart grids by enabling advanced analytics, predictive modeling, and autonomous decision making. AI algorithms can analyze vast amounts of data collected from sensors, meters, and other grid components to identify patterns, predict potential issues, and optimize grid operations in real time [88]. This synergy between 6G and AI revolutionizes the efficiency, reliability, and resilience of smart grids, paving the way for a more sustainable and intelligent energy infrastructure.

6.1.5. Brain-Computer Interfaces

Brain–computer interfaces (BCIs) aim to establish a direct link between the brain and a computer, enabling individuals to manipulate machines through their thoughts [89]. Unlike traditional input devices, such as a mouse or keyboard, a BCI decodes and interprets brain signals and converts them into control commands that the computer can execute. The

for instance, to operate a prosthetic limb or a wheelchair. With the emergence of 6G, BCIs could potentially benefit from advancements in communication technologies [90]. 6G networks are projected to offer higher data transfer rates and shorter latencies, making it possible to process brain signals in real time. This is crucial for the efficacy of BCIs, as real-time processing and analysis of brain signals are vital. The new technologies, such as terahertz communication and edge computing, available with 6G can potentially lead to the creation of advanced, compact BCI devices with improved precision and reliability. Furthermore, integrating BCIs with other cutting-edge technologies, such as IoT and AI, could offer new avenues for developing BCI applications, including context-aware and personalized BCI-based solutions. It is essential to note that although 6G holds excellent promise for BCIs, much research and development is still required to realize these possibilities fully.

6.1.6. Blockchain

Blockchain is a system that allows for the secure and transparent recording of digital transactions in a decentralized manner [91]. Integrating 6G networks with blockchain technology offers a range of benefits and potential applications. 6G, with its projected enhancements in data transfer rates, lower latency, and increased network capacity, could support the real-time processing of complex transactions and applications.

One potential use of blockchain technology in 6G is to improve security and privacy in communication and transactions [92]. The increased speed and capacity of 6G networks may enable the implementation of blockchain-based solutions to enhance data security and protect against cyberattacks and data breaches. Another application is the development of decentralized platforms and networks by integrating blockchain technology with 6G. 6G's enhanced speed and capacity could support the creation of decentralized networks that securely store and manage large amounts of sensitive data, such as financial transactions or medical records. These decentralized systems may offer improved security and privacy compared to centralized systems and could drive the development of new services and applications. Although 6G is still in its early stages of development, the potential applications of blockchain technology in 6G are vast and have the potential to significantly change the way we communicate, store, and manage data.

6.1.7. Space Travel

Space exploration is an area that could greatly benefit from the advancements in 6G technology. With its improved speed and capacity, 6G could facilitate real-time communication between spacecraft and ground control, streamlining missions and enabling agile decision making [93]. The fast data transfer speeds offered by 6G allow the effective transfer of substantial volumes of remote sensing data, leading to more accurate and detailed information and the potential for new scientific discoveries. Furthermore, the strong and secure communication networks enabled by 6G could connect spacecraft and ground control, ensuring reliable and uninterrupted communication. Additionally, 6G's enhanced network capacity and low latency could support the transmission of high-resolution virtual and augmented reality data, offering an improved immersive experience for those involved in space exploration.

6.1.8. Deep-Sea Sightseeing

Applying 6G in deep-sea sightseeing could enhance the underwater experience for individuals. With 6G's increased data transfer rates and reduced latency, real-time communication between the deep sea and the surface could be established [94]. This could allow for the transmission of high-quality images, videos, and data from the ocean's depths in real time, providing a more immersive experience for deep-sea observers. Additionally, 6G's improved network capacity and increased speed could support the deployment of underwater drones and other autonomous vehicles for deep-sea exploration [95]. These

vehicles could be equipped with high-resolution cameras and other sensing devices to collect and transmit data, enabling the collection of more accurate and detailed information about the deep-sea environment. The implementation of 6G in deep-sea sightseeing holds great promise, but much research and development work is still needed to realize its full potential.

6.1.9. Tactile Internet

The tactile internet is an emerging field that seeks to create a new form of humanmachine interaction through the sense of touch [96]. Applying 6G technology to the tactile internet could significantly enhance its capabilities and potential applications. 6G makes it possible for real-time, high-fidelity transmission of touch-based data. With 6G, it may be possible to create more advanced and responsive haptic systems that can provide a realistic simulation of touch, allowing for remote control and manipulation of objects, including virtual and augmented reality applications. 6G could also provide the highspeed and low-latency connectivity required to support the real-time teleoperation of robots and other remote-controlled devices, allowing for more precise and effective control. Additionally, 6G's advanced communication technologies, such as edge computing and terahertz communication, may enable the development of compact and highly accurate haptic devices [97]. Integrating 6G technology into the tactile internet could open up new possibilities for human-machine interaction and have far-reaching implications for healthcare, gaming, and manufacturing industries.

6.1.10. Industrial Internet of Things

The combination of 6G and the potential for transformation lies within the Industrial Internet of Things (IIoT) industrial operations [98]. 6G's real-time communication capabilities can result in more agile and efficient processes. Additionally, 6G's high data processing speeds can lead to improved decision making and increased accuracy in industrial processes. 6G's advanced security features can better protect against cyber threats and data breaches in industrial settings. Integrating 6G and IIoT can also open the door to new IoT-based solutions, such as the predictive maintenance and remote control of industrial systems [99]. With its ability to drive the creation of intelligent and automated industrial systems, 6G has the potential to increase productivity and efficiency and to lower costs.

6.1.11. Mixed and Augmented Reality

6G technology presents an incredible opportunity to enhance mixed and augmented reality (MAR) experiences. 6G's real-time capabilities enable the seamless merging of virtual and physical realms, providing users with a more immersive and interactive experience [100]. The transmission of high-resolution virtual and augmented reality data enabled by 6G can provide improved visual and sensory experiences in MAR applications. This opens up new possibilities for education, entertainment, and product visualization. Additionally, 6G's ability to connect individuals in virtual environments can lead to new ways for remote work, social interaction, and gaming to occur. 6G's enhanced network security measures can provide peace of mind, protecting sensitive information and user data from cyber threats [101]. The merging of 6G and MAR technology has significant potential to generate inventive and immersive experiences, establishing it as a primary application of 6G technology.

6.1.12. AI and Robotics

The application of 6G on AI and robotics is expected to be significant and impactful due to the increased capabilities and improved connectivity of 6G networks [102]. With 6G, AI algorithms will see a boost in accuracy and speed, while autonomous robots and drones will be equipped with real-time communication and control features. Advanced AI-powered systems, such as self-driving vehicles, smart factories, and intelligent homes, will become more sophisticated. With increased natural language processing abilities and

a more comprehensive range of applications, virtual assistants will also improve. 6G will enable the remote control and monitoring of AI and robotic systems in hazardous environments, and AI will be used for predictive maintenance and monitoring in industrial settings. The increased connectivity and capabilities of 6G networks will also drive the creation of new and innovative AI-powered applications and services.

6.1.13. Autonomous Vehicles and Smart Transportation Systems

The application of 6G technology in autonomous vehicles and smart transportation systems is poised to bring significant advancements and improvements [103]. 6G networks will offer the vital infrastructure for the secure and effective functioning of autonomous vehicles, facilitating real-time communication and control among vehicles, the central traffic management system, and the surrounding infrastructure [104]. The deployment of 6G will augment the safety and dependability of autonomous vehicles through more rapid and precise decision making. Furthermore, real-time data exchange between vehicles and infrastructure will optimize traffic management and flow, increasing efficiency and reducing congestion. The superior connectivity and features of 6G networks will foster the growth of cutting-edge smart transportation systems and services while also advancing existing autonomous vehicle technologies, such as sensors and mapping capabilities [105].

6.1.14. Mission-Critical Services (MCSs)

6G networks offer a transformative platform for enhancing mission-critical services (MCSs) through ultra-reliable low-latency communication, real-time monitoring and control of critical infrastructure, integration with edge computing and AI for predictive analytics and decision making, advanced public safety and emergency response applications, development of smart infrastructure and utilities for improved efficiency and resilience, support for telemedicine and remote healthcare services, and robust cybersecurity measures to protect sensitive data and infrastructure from cyber threats. These capabilities enable timely decision making, seamless coordination among emergency services, enhanced reliability of essential services, and improved accessibility to critical care, ultimately ensuring MCSs' reliability, responsiveness, and efficiency across various sectors [106].

6.1.15. Public Protection and Disaster Relief (PPDR)

The advent of 6G networks presents many transformative applications for public protection and disaster relief (PPDR) efforts. Through ultra-high-speed, low-latency communication, 6G facilitates the real-time data transmission essential for swift decision making and coordination among emergency responders during crises [107]. Integration with augmented reality (AR) and virtual reality (VR) technologies enhances situational awareness and navigation in disaster zones. At the same time, AI-powered predictive analytics enables proactive PPDR strategies by analyzing vast datasets to anticipate risks and optimize resource allocation. Seamless integration with drones and UAVs enables aerial surveillance and search-and-rescue missions, while biometric identification and wearable technologies ensure the safety and accountability of personnel in disaster areas. Advanced security measures safeguard sensitive PPDR data and communication infrastructure, while community engagement platforms empower citizens to participate actively in disaster preparedness and response efforts. Overall, 6G networks hold immense potential to enhance the effectiveness, efficiency, and resilience of PPDR initiatives, ultimately saving lives and mitigating the impact of disasters on communities [108].

7. Challenges for 6G Deployment

The deployment of 6G technology faces numerous deployment challenges. Some of them are discussed in this section (Figure 5).



Figure 5. Challenges for 6G deployment .

- **Technology innovation and standardization:** Technical difficulties are a challenge in implementing new enabling technologies like millimeter- and terahertz-wave communication, massive and ultra-massive MIMO, AI, ML, quantum communication, and ultra-reliable low-latency communication [109].
- **Bandwidth scarcity:** Identifying and allocating sufficient spectrum in the Terahertz (THz) frequency range for 6G is a significant challenge. THz frequencies offer the potential for high data rates but come with propagation challenges and require new regulatory frameworks [110].
- **Interoperability with existing networks:** Ensuring interoperability between different technologies across various industries and use cases is a complex challenge as many other networks use different standards and protocols [111].
- **Investment cost:** The deployment of 6G infrastructure is expected to be cost-intensive, requiring substantial investments in advanced technologies, equipment, and infrastructure. This might pose a financial challenge for network operators and end-users. This financial burden could hinder the broad adoption of 6G, especially in less economically developed regions and remote rural areas [112].
- Regulation and policy: Regulatory issues may arise due to new spectra and technologies, necessitating developing and implementing new policies and regulations [113].
- **Power consumption:** Power consumption is another concern, as the increased data rates and the number of devices connected to the network will result in higher power usage. Sharing a spectrum and infrastructure, implementing cell-free massive MIMO, and integrating communication and sensing are all pivotal aspects. Yet, the paramount transformation with 6G lies in the shift to higher frequencies, surpassing the 100 GHz threshold [114].
- International collaboration and harmonization: The competitive landscape, with multiple companies and countries vying to be the first to launch and deploy 6G, is a challenge. Promoting collaboration and harmonization of 6G standards and regulations on a global scale is crucial to ensure the success and widespread adoption of 6G technology and will be challenging.
- Security and privacy: There will be new security concerns as the network will transmit large amounts of sensitive data. Besides increasing connectivity and integrating various devices and systems, security and privacy will be a significant challenge [115].
- Environmental concerns: The production of 6G infrastructure requires various raw materials, including rare earth metals and minerals. The extraction processes can have environmental and social impacts, contributing to habitat destruction, pollution, and resource depletion [116].

Table 3 summarizes the 6G deployment challenges and possible solutions.

| Challenges | Possible Solutions |
|--|---|
| Technology innovation and standardization | Establish testbeds to validate the performance of millimeter- and terahertz-wave communication in different environments. This includes testing for signal propagation, interference, and device compatibility. Invest in developing signal-processing algorithms that can efficiently handle the massive number of antennas involved in MIMO systems. This includes beamforming, channel estimation, and interference management [117–120]. |
| Scarcity of high-frequency spectra for bandwidth allocation | Collaborate with regulatory bodies to identify and allocate specific frequency bands for 6G, focusing on millimeter and terahertz bands. This involves conducting spectrum studies to identify underutilized or unallocated frequency ranges [121,122]. |
| Create interoperability between current and 6G networks | The technologies should be built to interoperate with the existing network and devices [123,124]. |
| Investment cost | Implement a phased approach to 6G deployment, focusing on specific geographic areas, use cases, or network functionalities. This approach minimizes high upfront costs [125]. |
| Regulatory and policy challenges | Establish international agreements and collaborate with regulatory bodies to harmonize spectrum allocation for 6G. Encourage the development of dynamic spectrum-sharing technologies to optimize spectrum utilization [126]. |
| Power consumption | A model for optimizing power has been introduced for a 6G-enabled massive IoT network. The primary objective is to enhance overall system performance, providing energy-saving features. Through efficient power resource management, the model minimizes power overhead attributed to the extensive number of connected devices. The proposed network assessment includes analyzing the maximum allocated power and spectral efficiency under various network operations and distinct precoding schemes [127,128]. |
| International collaboration and harmonization | Encourage international collaboration in standardization bodies to develop unified standards for 6G technologies. Harmonize spectrum allocation, protocols, and interfaces to ensure interoperability and a consistent user experience [129]. |
| Security and privacy | While robust security mechanisms are in place for safeguarding data during transit, there is a pressing need to prioritize data protection in processing and storage for comprehensive end-to-end security in 6G. Techniques such as oblivious computing, confidential computing, homomorphic encryption, and privacy-centric identifiers can be employed across 6G network services and components [130,131]. |
| Environmental concerns | Design devices and infrastructure for longevity and ease of recycling. Establish collection and recycling programs for end-of-life electronic components. Encourage manufacturers to adopt sustainable product life cycles [132,133]. |
| The complexity of AI integration | Since AI algorithms require large amounts of data and computational resources, which can strain network infrastructure and increase operational overhead, develop AI algorithms optimized for resource-constrained environments to reduce computational complexity. Implement edge computing and distributed AI architectures to offload processing tasks from centralized networks and invest in AI hardware accelerators and efficient algorithms to improve performance while minimizing resource consumption [134]. |

Table 3. Summary of challenges and possible solutions for 6G deployment.

8. Key Technologies for 6G Deployment

Several key technologies are being explored and considered as potential components of 6G networks as shown in Figure 6. We will discuss a few major enabling technologies for 6G networks.

8.1. Terahertz Communication

Terahertz communication is expected to significantly impact the development of 6G networks by offering faster and more efficient data transmission capabilities [135]. Terahertz communication provides higher data rates than current wireless communication technologies, resulting in more immediate download and upload speeds and improving the overall user experience. Moreover, terahertz frequencies provide more available bandwidth, allowing for more efficient spectrum use, reducing network congestion, and improving overall network performance [136]. Terahertz communication can enable new use cases, such as high-resolution imaging, remote sensing, and advanced medical imaging, which require higher data rates and lower latency [137]. Using short wavelengths that are difficult



to intercept or detect, terahertz communication can enhance wireless communication security, reducing the risk of cyber attacks and unauthorized access [138].

Figure 6. Overview of key enabling technologies for 6G networks. This diagram illustrates the advanced technological pillars essential for the deployment of 6G networks, including quantum communication, beamforming, ultra-massive MIMO, reconfigurable intelligent surfaces, AI/ML, millimeter waves, UAV or satellite communication, and terahertz waves. Each technology is crucial for enhancing future wireless communication systems data rate, reliability, and overall efficiency.

8.2. Ultra-Massive MIMO

Ultra-massive MIMO technology, a key component in the evolution of 6G networks, offers significant advancements in network capacity, data rates, and coverage. It utilizes an extensive array of antennas capable of transmitting and receiving multiple data streams simultaneously. This capability not only accelerates data transmission but also significantly boosts data rates [139]. Furthermore, ultra-massive MIMO enhances signal processing and improves beamforming, leading to higher energy efficiency and more effective spectrum utilization.

This technology also plays a crucial role in optimizing frequency spectrum use, augmenting network capacity, and enhancing coverage. Its ability to reduce interference substantially improves overall network performance. Additionally, ultra-massive MIMO is instrumental in supporting emerging applications that demand higher data rates and lower latency. These applications include virtual reality, autonomous vehicles, and the development of smart city infrastructure [140]. Consequently, ultra-massive MIMO is a transformative technology poised to revolutionize 6G network capabilities and facilitate a new wave of technological advancements.

8.3. Beamforming

Beamforming is a crucial technology that improves the efficiency and reliability of wireless communication and enables the development of 6G networks. This technology involves directing radio waves in a specific direction to achieve better spectrum use and network performance. In 6G networks, beamforming can focus wireless signals on the desired receiver, resulting in reduced interference and improved signal strength [141]. This leads to higher data rates, lower latency, and the ability to support real-time data transmission for new use cases like virtual reality, remote surgery, and autonomous vehicles. Beamforming technology enables the more efficient use of the frequency spectrum by directing radio waves to specific areas [142]. This decreases interference and enhances network capacity, mitigating congestion and improving overall performance.

8.4. Cell-Free Massive MIMO

Cell-free massive MIMO is a promising technology for the development of 6G networks, as highlighted in [143]. This technology involves deploying numerous antennas across a given area, enhancing the efficiency of wireless communication. Compared to traditional cellular networks, cell-free massive MIMO offers several advantages, including improved network coverage. This is particularly beneficial in dense urban environments, where it allows for the more effective use of available radio resources, leading to better signal quality and fewer coverage gaps [144]. Cell-free massive MIMO technology leverages AI for resource allocation, interference coordination, and user scheduling to maximize network capacity and mitigate interference.

Additionally, cell-free massive MIMO can support a more significant number of users per unit area, thus increasing network capacity. This feature is especially useful in areas with high user density, such as stadiums, airports, and other public spaces [145]. Another significant advantage of cell-free massive MIMO is the reduction in latency. By enabling multiple users to access the same channel simultaneously, it reduces waiting times and improves the overall user experience [146].

Furthermore, cell-free massive MIMO contributes to enhanced energy efficiency. Reducing the need for complex and power-intensive signal processing algorithms leads to lower power consumption and extended battery life for mobile devices [147]. These benefits make Cell-free Massive MIMO a transformative technology for future cellular networks.

8.5. Millimeter Waves

Millimeter waves (mmWaves) operate within a frequency range of 30 GHz to 300 GHz and have shorter wavelengths than the traditional microwave bands used in 4G and 5G networks [148]. Their potential to deliver faster data speeds, higher network capacity, and improved network efficiency make them a crucial enabler of 6G networks. MmWave technology provides several benefits, such as enabling high data rates of several gigabits per second and allowing for new use cases, such as augmented reality and 8K video streaming, that require high data rates and low latency. mmWave can increase network capacity as the higher frequencies make more efficient use of the available spectrum. AI algorithms dynamically adjust transmission parameters and predict channel blockage events, ensuring reliable connectivity and maximizing data rates in millimeter-wave communication.

However, the use of mmWave technology also poses challenges. One such challenge is the shorter range of mmWave signals compared to traditional microwave frequencies, making it easy for obstacles such as buildings and trees to block signals and affect network coverage [149]. Furthermore, deploying mmWave technology requires many antennas, resulting in high infrastructure costs that must be addressed to promote widespread technology adoption.

8.6. Reconfigurable Intelligent Surfaces

Reconfigurable intelligent surfaces (RISs) comprise an emerging technology with significant potential to drive the development of 6G networks. Characterized by a flat surface embedded with numerous small antennas or reflectors, a RIS can electronically control radio waves to reflect, amplify, or absorb them [150]. This capability offers several advantages for future network infrastructures.

One of the primary benefits of RIS technology is its ability to enhance network coverage. Placing RISs in areas with traditionally poor coverage or high signal interference, such as indoor spaces, can effectively reflect and amplify signals. This improvement results in stronger signal strength and broader coverage. Additionally, RIS technology contributes to increased network capacity by enabling more efficient use of available radio resources. It accomplishes this by focusing radio waves in specific directions, reducing interference and supporting more concurrent users.

Reconfigurable intelligent surfaces (RISs) are optimized by AI algorithms for efficient signal reflection and amplification, enhancing network coverage and capacity. AI-driven surface configurations adapt to changing environmental conditions and user demands, optimizing energy efficiency and spectral utilization for improved network performance.

Another significant advantage of RIS technology is its contribution to energy efficiency. By reflecting and focusing radio waves directionally, a RIS minimizes the energy required for transmitting signals over long distances [151]. This leads to lower power consumption, prolonged battery life for mobile devices, and a reduction in the overall energy footprint of the network. The widespread implementation of RIS technology will necessitate substantial investments in infrastructure and technological advancements. Addressing these requirements is essential to fully leverage the capabilities of RISs in enhancing future network systems.

8.7. Quantum Communication

Quantum communication is an advanced technology that can be utilized in the development of 6G networks [152]. Unlike traditional communication technologies that rely on electromagnetic waves to transmit data, quantum communication uses photons for transmission. This feature allows quantum communication to offer high levels of security, making it ideal for military and government communications where the highest levels of security are required.

Quantum communication can also support new applications that necessitate real-time data transmission, such as autonomous vehicles and smart cities. By offering instantaneous communication over long distances, quantum communication can reduce latency and enable faster response times [153]. However, deploying this technology faces several challenges, such as the need for specialized hardware and infrastructure and high implementation costs.

8.8. UAV/Satellite Communication

UAV/satellite communication is a technology that can facilitate the development of 6G networks [154]. This technology uses unmanned aerial vehicles (UAVs) and satellites to provide wireless connectivity to remote and underserved areas. By using these aerial platforms, it is possible to provide high-speed data transfer and internet connectivity to regions that are difficult to reach using traditional terrestrial networks. There is potential for expanded network coverage, particularly in remote and rural areas with limited conventional infrastructure. This can enable more people to access high-speed internet and other data services, improving access to information and enabling new applications and services. UAV/satellite communication can also support new use cases that require real-time data transmission, such as remote medical procedures and disaster response [155]. By enabling communication over long distances and in rugged terrain, UAV/satellite communication can improve the efficiency and effectiveness of these applications. The deployment of UAV/satellite communication technology also presents challenges, such as the need for specialized hardware and infrastructure, as well as regulatory issues related to the use of airspace [156]. The technology requires significant satellite and UAV deployment and maintenance investment.

Table 4 shows the role of AI in the above-discussed key technologies for 6G deployment.

| Key Technologies for 6G Deployment | The Role of AI and ML in the Key Technologies |
|-------------------------------------|--|
| Terahertz communication | Terahertz communication in 6G networks promises faster and more efficient data transmission, aided by AI's ability to optimize signal processing and security measures. Through real-time adaptation of transmission parameters, AI enhances network performance and mitigates signal attenuation, ensuring robust and secure wireless communication [157]. |
| Ultra-massive MIMO | AI algorithms dynamically adjust antenna configurations and predict user demand patterns, maximizing spectral efficiency and user throughput. This synergy between AI and ultra-massive MIMO enhances network coverage and reliability while mitigating interference for improved quality of service [158]. |
| Beamforming | AI-driven beamforming techniques enhance spectral efficiency and mitigate multi-path fading, ensuring seamless connectivity and reliability in dynamic urban environments [159]. |
| Cell-free massive MIMO | Cell-free massive MIMO, characterized by deploying numerous antennas across an area to enhance wireless communication efficiency, leverage AI for various crucial tasks. AI algorithms are utilized for resource allocation, interference coordination, and user scheduling to maximize network capacity and mitigate interference effectively [160]. |
| Millimeter waves | AI algorithms dynamically adjust transmission parameters and predict channel blockage events, ensuring reliable connectivity and maximizing data rates in millimeter-wave communication [161]. |
| Reconfigurable intelligent surfaces | The role of AI in reconfigurable intelligent surfaces (RISs) is instrumental in optimizing their configuration for efficient signal reflection and amplification, thereby enhancing network coverage and capacity. AI algorithms adaptively adjust RIS configurations based on changing environmental conditions and user demands, optimizing energy efficiency and spectral utilization for improved network performance [162]. |
| Quantum communication | Quantum communication in 6G networks is enhanced by AI's optimization of quantum key distribution protocols and network management systems. AI algorithms optimize encoding strategies and automate key distribution tasks, ensuring secure and reliable data transmission over quantum communication channels [163]. |
| UAV/satellite communication | UAV/satellite communication benefits from AI-driven optimization of trajectories and resource allocation, ensuring seamless connectivity in remote areas by predicting user demand patterns and dynamically allocating resources, enhancing network coverage and reliability for improved access to high-speed internet and data services [164]. |

Table 4. The role of AI in the key technologies for 6G deployment.

9. Is 6G Dangerous for Your Health?

Research on the potential health effects of 6G networks is limited, as the technology is still in its early stages of development. However, concerns have been raised about the possible risks of exposure to high-frequency electromagnetic radiation in 6G networks [165]. The World Health Organization has categorized electromagnetic radiation as a potential carcinogen, and specific research studies have associated exposure to high levels of electromagnetic radiation with an elevated likelihood of cancer and other health issues [166]. However, these studies have focused mainly on exposure to radiofrequency radiation from cell phones and other devices that operate in lower-frequency ranges used by 4G and 5G networks [167]; while there is currently no evidence to suggest that exposure to the higher-frequency electromagnetic radiation used in 6G networks poses a significant health risk to humans, more research is needed to fully understand this technology's potential health effects. According to the FCC, the frequency range designated for 6G is between 95 GHz to 3THz. Despite being three to a thousand times higher than 5G's frequency, these ranges are still considered safe as they are non-ionizing [168].

Exposure to electromagnetic radiation from 6G networks is likely to be significantly lower than that from other sources, as the technology will likely use a combination of different frequency ranges, including lower frequencies employed in earlier generations of mobile networks. Measures can also be taken to reduce exposure to electromagnetic radiation, for instance, restricting cell phone and wireless device usage, employing protective cases, and keeping devices away from the body when in use [169]; while there is currently no evidence to suggest that 6G networks are dangerous for human health, further research is needed to understand the potential risks of this technology.

10. Recommendations and Future Research Direction

- To support the establishment of dedicated testbeds to validate and maximize the performance of millimeter- and terahertz-wave communication in diverse environments, reinforcement learning algorithms can be deployed to optimize the configuration of testbed parameters like antenna placement, transmit power, and channel allocation. Further research is needed on new schemes adapted for the high-frequency bands and extensive bandwidths of 6G. This includes studying techniques for enhanced spectrum utilization and improved data throughput, critical for reliable communication in various environments, and training ML algorithms to process and analyze signals received from millimeter- and terahertz-wave communication systems. These algorithms can adaptively adjust parameters to optimize signal quality and mitigate interference in diverse environments.
- There is a need to design smart and intelligent protocols and architectures for efficient network handover and connectivity in remote areas exhibiting adaptability by seamlessly adjusting to changing network conditions, such as signal strength fluctuations and congestion. These systems should demonstrate context awareness by gathering and analyzing contextual data like location and user preferences to inform decision making. These systems should utilize predictive algorithms to anticipate handover events, enabling proactive connectivity and resource allocation optimization. With dynamic resource allocation, they can optimize network performance and efficiency by allocating bandwidth and spectrum based on real-time demands. Efficient routing algorithms should be utilized to minimize latency and maximize throughput during handover and connectivity establishment with intelligent decision-making mechanisms, including ML algorithms that can autonomously analyze network conditions to make optimized decisions regarding handover initiation and resource allocation.
- Natural language processing techniques can process unstructured network data, such as log files and network performance reports, to extract insights and identify potential congestion points or anomalies. AI-driven congestion management will dynamically monitor network traffic and identify congestion points, implementing strategies such as traffic shaping and rerouting to alleviate congestion and maintain optimal performance. Through predictive analytics, AI algorithms analyze historical network data to forecast future network behavior, enabling proactive resource allocation and capacity planning to prevent congestion and service degradation.
- Improvements to address the challenges posed by the increasing number of connected devices entail a two-fold approach: creating low-power hardware solutions and implementing sustainable network operation methods in the 6G network. Low-power hardware solutions involve designing energy-efficient components, optimizing circuit designs, employing low-power communication protocols, and integrating energy harvesting technologies like kinetic energy harvesters, piezoelectric materials, radio frequency energy harvesting, and thermoelectric generators into devices. Sustainable network operation methods encompass dynamic power management, resource sharing and allocation, deployment of green networking technologies, development of energy-aware routing algorithms, and implementation of life-cycle management practices.
- Invest in research and development of AI-driven encryption techniques, such as homomorphic encryption, which is a form of encryption that allows computations to be performed on encrypted data without decrypting them first. In other words, it enables operations such as addition and multiplication to be carried out on ciphertexts, resulting in the same operations being performed on the plaintexts when decrypted. This property is beneficial in scenarios where data privacy is crucial, as it allows for secure computation of sensitive data while keeping them encrypted throughout the process. For example, lattice-based cryptography utilizes the hardness of specific lattice problems, such as the shortest vector problem (SVP) or the learning with errors (LWE) problem, as the basis for cryptographic schemes. These problems are believed

to be computationally difficult to solve, even for quantum computers, making latticebased cryptography resistant to attacks from both classical and quantum adversaries. Post-quantum cryptography utilizes the hardness of specific lattice problems, such as the shortest vector problem (SVP) or the learning with errors (LWE) problem, as the basis for cryptographic schemes. These problems are believed to be computationally difficult to solve, even for quantum computers, making lattice-based cryptography resistant to attacks from classical and quantum adversaries. These techniques leverage AI algorithms to enhance data encryption security, enabling secure storage and transmission of sensitive information in interconnected environments, and can be improved with more advanced research.

- Research should focus on developing quantum key distribution schemes that are practical, scalable, and compatible with the high-speed and high-bandwidth requirements of 6G networks. Further investigation is needed into using entanglement as a resource for secure and efficient data transmission in 6G networks. Researchers should explore methods for generating, distributing, and utilizing entangled quantum states to enhance communication security and enable novel communication protocols with superior performance.
- Research in 6G networks should focus on leveraging their unique capabilities, including ultra-high data rates, ultra-reliable low-latency communication (URLLC), massive connectivity, high-precision positioning and navigation, edge computing, and network slicing, to enable transformative applications such as immersive virtual and augmented reality (VR/AR), autonomous vehicles, smart city infrastructure, digital twins, and edge computing. By exploring the potential of these capabilities and developing innovative solutions that harness their full potential, researchers can pave the way for a more connected, intelligent, and efficient future, where advanced applications and services enhance user experiences and drive societal and economic benefits.
- Researchers should focus on developing advanced network slicing in 6G networks that encompass architectures, algorithms, and management frameworks that enable the customization and efficient allocation of network resources for different services and applications. Hierarchical slicing architectures hierarchically organize network slices to facilitate resource sharing and isolation, while cloud-native slicing architectures leverage cloud-native principles for dynamic slice creation and deployment. Dynamic resource allocation algorithms dynamically allocate network resources among slices based on real-time demands, while QoS-aware routing algorithms determine optimal paths for traffic within slices. SDN-based and NFV-based slicing management frameworks facilitate centralized control and automated life-cycle management of network slices, enabling operators to efficiently tailor network services to diverse application requirements.
- Focusing on ultra-reliable low-latency communication (URLLC) in 6G networks is crucial for supporting critical applications such as remote healthcare and industrial automation. Ultra-reliable low-latency communication (URLLC) in 6G networks relies on specific standards, protocols, and technologies tailored to meet the stringent requirements of critical applications like remote healthcare and industrial automation. The 5G NR URLLC standard defined by 3GPP specifies parameters for achieving ultra-reliable and low-latency communication [170]. At the same time, IEEE 802.1 Time-Sensitive Networking (TSN) provides mechanisms for deterministic and low-latency communication in industrial automation. One-Pair Ethernet (OPE) offers cost-effective communication for automotive applications, and Time-Critical Networking (TCN) standards by the Industrial Internet Consortium ensure deterministic communication in industrial settings. Additionally, Non-Orthogonal Multiple Access (NOMA) is explored as a promising technique for efficient and reliable URLLC in dense networks. These standards, protocols, and technologies collectively enable ultra-reliable and low-latency communication, crucial for supporting critical services in 6G networks.

11. 7G Networks

7G networks, currently a theoretical concept, are anticipated to significantly outpace the capabilities of 6G networks, introducing groundbreaking features such as holographic communication and brain–computer interfaces. Presently, the term "7G" has not been officially recognized by any standardization bodies, and the realization of these networks is projected to span several decades. This timeline is due to the substantial advancements needed in wireless communication technology and infrastructure.

The vision for 7G includes the potential use of beyond-terahertz frequencies, which could dramatically increase data transmission speeds. Additionally, advancements in neuromorphic computing are expected to enhance data processing efficiency substantially. Among the other prospective features of 7G networks are highly sophisticated AI, seamless inter-network connectivity, and novel applications like fully autonomous transportation systems.

Despite the exciting prospects that 7G networks present, their development poses significant challenges. These include the need for extensive investment in research and development and considerable upgrades to existing communication infrastructure. Such developments are essential to leap from theoretical ideas to practical implementation, and this process is likely to span multiple decades.

12. Conclusions

This paper provides an in-depth exploration of the evolving landscape of 6G networks, highlighting the transformative role of artificial intelligence in this next wave of wireless technology. We presented the hierarchical structure of 6G networks, scrutinizing the technological advancements driving its emergence, the potential applications, and the challenges it faces. We explored the evolutionary trajectory of wireless communication technologies, emphasizing how each generation has progressively enhanced data rates, connectivity, and functionality. With the advent of 6G, we anticipate an unprecedented leap in these aspects, driven by key enabling technologies like terahertz communication, ultramassive MIMO, AI, ML, quantum communication, and reconfigurable intelligent surfaces.

Integrating AI into 6G networks emerged as a pivotal theme, promising to revolutionize network management and user experience. By leveraging AI's predictive and adaptive capabilities, 6G networks are poised to offer optimized bandwidth, enhanced efficiency, and more personalized services. The potential applications of 6G are vast and varied, ranging from enhanced mobile broadband to innovative domains such as smart cities, autonomous systems, and brain–computer interfaces, underlining the network's transformative impact across sectors.

However, the journey towards actualizing 6G is not without its challenges. Technological hurdles, bandwidth scarcity, interoperability issues, investment costs, regulatory complexities, environmental concerns, and security and privacy issues represent significant barriers. Addressing these challenges requires a concerted effort from industry, academia, and regulatory bodies. As we venture into the speculative realm of 7G networks, we contemplate even more advanced features and capabilities. The prospect of holographic communication and neuromorphic computing in 7G underscores wireless technology's continual evolution and boundless potential.

In conclusion, this paper serves as a comprehensive overview of 6G networks and a catalyst for further research and development. It underlines the synergy between 6G and AI technologies and sets the stage for continued exploration and innovation. As the world gravitates towards increasingly connected and intelligent systems, the insights and discussions presented here will be instrumental in shaping the future of wireless communication.

Author Contributions: The authors declare that they have equally contributed to the paper. All authors read and approved the final manuscript.

Funding: This research received no external funding.

Acknowledgments: We want to thank the authors of the literature cited in this paper for contributing useful ideas to this study.

Conflicts of Interest: The authors declare no conflicts of interest. The funders had no role in the design of the study; in the collection, analyses, or interpretation of data; in the writing of the manuscript, or in the decision to publish the results.

Abbreviations

The following abbreviations are used in this manuscript:

| MDPI | Multidisciplinary Digital Publishing Institute |
|-------|--|
| DOAJ | Directory of open access journals |
| TLA | Three-letter acronym |
| LD | Linear dichroism |
| AMPS | Advanced Mobile Phone System |
| TACS | Total Access Communication System |
| NMT | Nordic Mobile Telephone |
| SMS | Short message service |
| PDC | Personal Digital Cellular |
| WCDMA | Wideband Code Division Multiple Access |
| MMS | Multimedia message support |
| RL | Reinforcement learning |
| RIS | Reconfigurable intelligent surface |
| | |

References

- 1. IMT Traffic Estimates for the Years 2020 to 2030. International Telecommunication Union (ITU). Available online: https://www.itu.int/pub/r-rep-m.2370 (accessed on 25 October 2023).
- 2. Bangerter, B.; Talwar, S.; Arefi, R.; Stewart, K. Networks and Devices for the 5G Era. *IEEE Commun. Mag.* 2014, 52, 90–96. [CrossRef]
- 3. Sinclair, M.; Maadi, S.; Zhao, Q.; Hong, J.; Ghermandi, A.; Bailey, N. Assessing the Socio-Demographic Representativeness of Mobile Phone Application Data. *Appl. Geogr.* 2023, *158*, 102997. [CrossRef]
- 4. Huseien, G.F.; Shah, K.W. A Review on 5G Technology for Smart Energy Management and Smart Buildings in Singapore. *Energy AI* 2022, 7, 100116. [CrossRef]
- Baier, P.; Dürr, F.; Rothermel, K. TOMP: Opportunistic Traffic Offloading Using Movement Predictions. In Proceedings of the 37th Annual IEEE Conference on Local Computer Networks, Clearwater Beach, FL, USA, 22–25 October 2012; pp. 50–58. [CrossRef]
- Gohar, A.; Nencioni, G. The Role of 5G Technologies in a Smart City: The Case for Intelligent Transportation System. *Sustainability* 2021, 13, 5188. [CrossRef]
- Tataria, H.; Shafi, M.; Molisch, A.F.; Dohler, M.; Sjöland, H.; Tufvesson, F. 6G Wireless Systems: Vision, Requirements, Challenges, Insights, and Opportunities. *Proc. IEEE* 2021, 109, 1166–1199. [CrossRef]
- 8. Murroni, M.; Anedda, M.; Fadda, M.; Ruiu, P.; Popescu, V.; Zaharia, C.; Giusto, D. 6G—Enabling the New Smart City: A Survey. *Sensors* 2023, 23, 7528. [CrossRef] [PubMed]
- Singh, P.R.; Singh, V.K.; Yadav, R.; Chaurasia, S.N. 6G Networks for Artificial Intelligence-Enabled Smart Cities Applications: A Scoping Review. *Telemat. Inform. Rep.* 2023, 9, 100044. [CrossRef]
- 10. AI-Powered 6G Networks Will Reshape Digital Interactions. Available online: https://www.technologyreview.com/2023/10/26 /1082028/ai-powered-6g-networks-will-reshape-digital-interactions/ (accessed on 29 October 2023).
- 11. Quy, V.K.; Chehri, A.; Quy, N.M.; Han, N.D.; Ban, N.T. Innovative Trends in the 6G Era: A Comprehensive Survey of Architecture, Applications, Technologies, and Challenges. *IEEE Access* **2023**, *11*, 39824–39844. [CrossRef]
- Shahen Shah, A.F.M. A Survey From 1G to 5G Including the Advent of 6G: Architectures, Multiple Access Techniques, and Emerging Technologies. In Proceedings of the 2022 IEEE 12th Annual Computing and Communication Workshop and Conference (CCWC), Las Vegas, NV, USA, 26–29 January 2022; pp. 1117–1123. [CrossRef]
- Chataut, R.; Akl, R. Massive MIMO Systems for 5G and Beyond Networks—Overview, Recent Trends, Challenges, and Future Research Direction. Sensors 2020, 20, 2753. [CrossRef] [PubMed]
- 14. Serôdio, C.; Cunha, J.; Candela, G.; Rodriguez, S.; Sousa, X.R.; Branco, F. The 6G Ecosystem as Support for IOE and Private Networks: Vision, Requirements, and Challenges. *Future Internet* **2023**, *15*, 348. [CrossRef]
- 15. Burbank, J.L.; Andrusenko, J.; Everett, J.S.; Kasch, W.T.M. Second-Generation (2G) Cellular Communications. In *Wireless Networking: Understanding Internetworking Challenges*; IEEE: New York, NY, USA, 2013; pp. 250–365. [CrossRef]
- Techopedia. (n.d.-a). Interim Standard-95 (IS-95). Available online: https://www.techopedia.com/definition/24248/interimstandard-95-is-95 (accessed on 13 March 2023).

- 17. Moore, K. From 1G to 5G: The Evolution of Mobile Communications. Mpirical, 11 August 2023. Available online: https://www.mpirical.com/blog/the-evolution-of-mobile-communication (accessed on 30 October 2023).
- Garg, V.K.; Halpern, S.; Smolik, K.F. Third Generation (3G) Mobile Communications Systems. In Proceedings of the 1999 IEEE International Conference on Personal Wireless Communications (Cat. No.99TH8366), Jaipur, India, 17–19 February 1999; pp. 39–43. [CrossRef]
- 19. Central Intelligence Agency. Available online: https://www.cia.gov/the-world-factbook/field/telecommunication-systems (accessed on 30 October 2023).
- Vergados, D.D.; Panoutsakopoulos, A.; Douligeris, C. Group Registration with Distributed Databases for Location Tracking in 3G Wireless Networks. *Comput. Netw.* 2008, 52, 1521–1544. [CrossRef]
- Dahlman, E.; Parkvall, S.; Sköld, J. Introduction. In 4G, LTE Evolution and the Road to 5G; Academic Press: Cambridge, MA, USA, 2016; pp. 1–5. [CrossRef]
- 22. Lopa, V. Evolution of Mobile Generation Technology: 1G to 5G and Review of Upcoming Wireless Technology 5G. Int. J. Mod. Trends Eng. Res. 2015, 2, 281–290.
- Deepender, M.; Shrivastava, U.; Verma, J.K. A Study on 5G Technology and Its Applications in Telecommunications. In Proceedings of the 2021 International Conference on Computational Performance Evaluation (ComPE), Shillong, India, 1–3 December 2021; pp. 365–371. [CrossRef]
- 24. Guevara, L.; Auat Cheein, F. The Role of 5G Technologies: Challenges in Smart Cities and Intelligent Transportation Systems. *Sustainability* **2020**, *12*, 6469. [CrossRef]
- 25. Nasralla, M.M.; Khattak, S.B.; Ur Rehman, I.; Iqbal, M. Exploring the Role of 6G Technology in Enhancing Quality of Experience for M-Health Multimedia Applications: A Comprehensive Survey. *Sensors* **2023**, *23*, 5882. [CrossRef] [PubMed]
- Kaiser, M.S.; Zenia, N.; Tabassum, F.; Mamun, S.A.; Rahman, M.A.; Islam, M.d.S.; Mahmud, M. 6G Access Network for Intelligent Internet of Healthcare Things: Opportunity, Challenges, and Research Directions. *Adv. Intell. Syst. Comput.* 2020, 1309, 317–328. [CrossRef]
- 27. Baker, S.; Xiang, W. Artificial Intelligence of Things for Smarter Healthcare: A Survey of Advancements, Challenges, and Opportunities. *IEEE Commun. Surv. Tutor.* 2023, 25, 1261–1293. [CrossRef]
- Kumar, R.; Gupta, S.K.; Wang, H.-C.; Kumari, C.S.; Korlam, S.S.V.P. From Efficiency to Sustainability: Exploring the Potential of 6G for a Greener Future. Sustainability 2023, 15, 16387. [CrossRef]
- 29. Nankya, M.; Chataut, R.; Akl, R. Securing Industrial Control Systems: Components, Cyber Threats, and Machine Learning-Driven Defense Strategies. *Sensors* 2023, 23, 8840. [CrossRef]
- 30. Ray, P.P. A perspective on 6G: Requirement, technology, enablers, challenges and future road map. *J. Syst. Archit.* **2021**, *118*, 102180. [CrossRef]
- 31. Ismail, L.; Buyya, R. Artificial Intelligence Applications and Self-Learning 6G Networks for Smart Cities Digital Ecosystems: Taxonomy, Challenges, and Future Directions. *Sensors* **2022**, *22*, 5750. [CrossRef]
- Kimachia, K.; Staff, T.; Clarke, M.; McQuarrie, K.; Millares, L.; Azhar, A.; Abbott, B. 5G vs 6G: What's the Difference? TechRepublic, 23 Febeuary 2023. Available online: https://www.techrepublic.com/article/5g-vs-6g/ (accessed on 1 November 2023).
- Adhikari, M.; Hazra, A. 6G-Enabled Ultra-Reliable Low-Latency Communication in Edge Networks. *IEEE Commun. Stand. Mag.* 2022, 6, 67–74. [CrossRef]
- Hokazono, Y.; Kohara, H.; Kishiyama, Y.; Asai, T. Extreme Coverage Extension in 6G: Cooperative Non-terrestrial Network Architecture Integrating Terrestrial Networks. In Proceedings of the 2022 IEEE Wireless Communications and Networking Conference (WCNC), Austin, TX, USA, 10–13 April 2022; pp. 138–143. [CrossRef]
- 35. Ruth, C. How 6G Can Transform the World and Technology. IEEE Standards Association. 2023. Available online: https://standards.ieee.org/beyond-standards/how-6g-can-transform-the-world-and-technology/ (accessed on 14 March 2024).
- 36. Jain, P.; Gupta, A.; Kumar, N.; Guizani, M. Dynamic and Efficient Spectrum Utilization for 6G with THz, mmWave, and RF Band. *IEEE Trans. Veh. Technol.* **2023**, *72*, 3264–3273. [CrossRef]
- 6G Flagship. Key Drivers and Research Challenges for 6G Ubiquitous Wireless Intelligence. Available online: https://www.6gflagship.com/key-drivers-and-research-challenges-for-6g-ubiquitous-wireless-intelligence/ (accessed on 2 November 2023).
- Ansere, J.A.; Kamal, M.; Khan, I.A.; Aman, M.N. Dynamic Resource Optimization for Energy-Efficient 6G-IoT Ecosystems. Sensors 2023, 23, 4711. [CrossRef]
- 39. Zhang, L.; Du, Q.; Lu, L.; Zhang, S. Overview of the Integration of Communications, Sensing, Computing, and Storage as Enabling Technologies for the Metaverse over 6G Networks. *Electronics* **2023**, *12*, 3651. [CrossRef]
- Shen, F.; Shi, H.; Yang, Y. A Comprehensive Study of 5G and 6G Networks. In Proceedings of the 2021 International Conference on Wireless Communications and Smart Grid (ICWCSG), Hangzhou, China, 13–15 August 2021; pp. 321–326. [CrossRef]
- Techopedia. When 6G Met AI: How Next-Gen Mobile Networks Will Work. Available online: https://www.techopedia.com/6gand-ai-next-gen-mobile-networks-will-change-the-world (accessed on 8 November 2023).
- 42. Ashwin, M.; Alqahtani, A.S.; Mubarakali, A.; Sivakumar, B. Efficient Resource Management in 6G Communication Networks Using Hybrid Quantum Deep Learning Model. *Comput. Electr. Eng.* **2023**, *106*, 108565. [CrossRef]
- Cloudait. Ai-Enabled Self-Healing Networks: A Crucial Pillar of 6G Reliability. AI Tools Practical Handbook. 1 August 2023. Available online: https://www.cloudaitech.net/ai-enabled-self-healing-networks-a-crucial-pillar-of-6g-reliability/ (accessed on 8 November 2023).

- 44. Saeed, M.M.; Saeed, R.A.; Abdelhaq, M.; Alsaqour, R.; Hasan, M.K.; Mokhtar, R.A. Anomaly Detection in 6G Networks Using Machine Learning Methods. *Electronics* **2023**, *12*, 3300. [CrossRef]
- 45. Al-Dulaimi, O.; Al-Dulaimi, M.; Al-Dulaimi, A.; Alexandra, M.O. Cognitive Radio Network Technology for IoT-Enabled Devices. *Eng. Proc.* **2023**, *41*, 7. [CrossRef]
- 46. Zhou, Y.; Liu, L.; Wang, L.; Hui, N.; Cui, X.; Wu, J.; Peng, Y.; Qi, Y.; Xing, C. Service-aware 6G: An Intelligent and Open Network Based on the Convergence of Communication, Computing, and Caching. *Digit. Commun. Netw.* **2020**, *6*, 253–260. [CrossRef]
- 47. Iqbal, A.; Tham, M.-L.; Wong, Y.J.; Al-Habashna, A.; Wainer, G.; Zhu, Y.X.; Dagiuklas, T. Empowering Non-Terrestrial Networks with Artificial Intelligence: A Survey. *IEEE Access* 2023, *11*, 100986–101006. [CrossRef]
- Ahammadi, A.; Hassan, W.H.; Shamsan, Z.A. An Overview of Artificial Intelligence for 5G/6G Wireless Networks Security. In Proceedings of the 2022 International Conference on Cyber Resilience (ICCR), Dubai, United Arab Emirates, 6–7 October 2022; pp. 1–6. [CrossRef]
- Dogra, A.; Jha, R.K.; Jha, K.R. Reinforcement Learning (RL) for Optimal Power Allocation in 6G Network. In Proceedings of the 2022 OPJU International Technology Conference on Emerging Technologies for Sustainable Development (OTCON), Raigarh, Chhattisgarh, India, 8–10 February 2023; pp. 1–6. [CrossRef]
- 50. Saraswat, S.K.; Deolia, V.K.; Shukla, A. Allocation of Power in NOMA-Based 6G-Enabled Internet of Things Using Multi-Objective Based Genetic Algorithm. J. Elect. Eng. 2023, 74, 95–101. [CrossRef]
- 51. Zou, S.; Wu, J.; Yu, H.; Wang, W.; Huang, L.; Ni, W.; Liu, Y. Efficiency-Optimized 6G: A Virtual Network Resource Orchestration Strategy by Enhanced Particle Swarm Optimization. *Digit. Commun. Netw.* 2023, *in press.* [CrossRef]
- 52. Chen, C.; Zhang, H.; Hou, J.; Zhang, Y.; Zhang, H.; Dai, J.; Pang, S.; Wang, C. Deep Learning in the Ubiquitous Human–Computer Interactive 6G Era: Applications, Principles and Prospects. *Biomimetics* **2023**, *8*, 343. [CrossRef]
- 53. de la Rosa, E.; Yu, W. Data-Driven Fuzzy Modeling Using Restricted Boltzmann Machines and Probability Theory. *IEEE Trans. Syst. Man Cybern. Syst.* **2020**, *50*, 2316–2326. [CrossRef]
- 54. Moubayed, A.; Shami, A.; Al-Dulaimi, A. On End-to-End Intelligent Automation of 6G Networks. *Future Internet* **2022**, *14*, 165. [CrossRef]
- 55. Chen, T.; Deng, J.; Tang, Q.; Liu, G. Optimization of Quality of AI Service in 6G Native AI Wireless Networks. *Electronics* 2023, 12, 3306. [CrossRef]
- 56. 5G/6G Wireless Networks. 5G/6G | Homeland Security. Available online: https://www.dhs.gov/science-and-technology/5g6g (accessed on 2 November 2023).
- 57. "Ubiquitous 6G: China Already Head Start?" INTEGRAL Website (EN). Available online: https://www.integralnewenergy.com/ ?p=31920 (accessed on 2 November 2023).
- 58. "Huawei Started Research on 6G 'A Long Time Ago', CEO Says". Available online: https://www.rcrwireless.com/20190930/5g/ huawei-started-research-6g-long-time-ago-ceo-says (accessed on 2 November 2023).
- 59. Times, G. China Ramping up Research into 6G. Global Times. Available online: https://www.globaltimes.cn/content/1188617. shtml (accessed on 2 November 2023).
- 60. Tomás, J.P. LG to Focus on 6G Research with Partnership with Keysight Technologies. RCR Wireless News, 8 April 2021. Available online: https://www.rcrwireless.com/20210408/business/lg-to-focus-on-6g-research-with-partnership-with-keysighttechnologies (accessed on 2 November 2023).
- 61. Jiang, W.; Han, B.; Habibi, M.A.; Schotten, H.D. The Road towards 6G: A Comprehensive Survey. *IEEE Open J. Commun. Soc.* **2021**, *2*, 334–366. [CrossRef]
- 62. O'Hara, J.F.; Ekin, S.; Choi, W.; Song, I. A Perspective on Terahertz Next-Generation Wireless Communications. *Technologies* **2019**, 7, 43. [CrossRef]
- 63. Writer, S. Japan Readies \$2bn to Support Industry Research on 6G Tech. Nikkei Asia, 21 November 2019. Available online: https://asia.nikkei.com/Business/Technology/Japan-readies-2bn-to-support-industry-research-on-6G-tech (accessed on 2 November 2023).
- 64. 6G on the Horizon: A Global Overview of the Latest Developments in Wireless Technology: Market Research Blog. Market Research Reports[®] Inc., 5 April 2023. Available online: https://www.marketresearchreports.com/blog/2023/04/05/6g-horizon-global-overview-latest-developments-wireless-technology (accessed on 2 November 2023).
- 65. Puspitasari, A.A.; An, T.T.; Alsharif, M.H.; Lee, B.M. Emerging Technologies for 6G Communication Networks: Machine Learning Approaches. *Sensors* 2023, 23, 7709. [CrossRef] [PubMed]
- 66. Banafaa, M.; Shayea, I.; Din, J.; Azmi, M.H.; Alashbi, A.; Daradkeh, Y.I.; Alhammadi, A. 6G Mobile Communication Technology: Requirements, Targets, Applications, Challenges, Advantages, and Opportunities. *Alex. Eng. J.* **2023**, *64*, 245–274. [CrossRef]
- Alsabah, M.; Naser, M.A.; Mahmmod, B.M.; Abdulhussain, S.H.; Eissa, M.R.; Al-Baidhani, A.; Noordin, N.K.; Sait, S.M.; Al-Utaibi, K.A.; Hashim, F. 6G Wireless Communications Networks: A Comprehensive Survey. *IEEE Access* 2021, *9*, 148191–148243. [CrossRef]
- 68. Ahmad, I.; Rodriguez, F.; Huusko, J.; Seppänen, K. On the Dependability of 6G Networks. Electronics 2023, 12, 1472. [CrossRef]
- 69. Kanellopoulos, D.; Sharma, V.K.; Panagiotakopoulos, T.; Kameas, A. Networking Architectures and Protocols for IoT Applications in Smart Cities: Recent Developments and Perspectives. *Electronics* **2023**, *12*, 2490. [CrossRef]
- Hazarika, A.; Rahmati, M. Towards an Evolved Immersive Experience: Exploring 5G- and Beyond-Enabled Ultra-Low-Latency Communications for Augmented and Virtual Reality. *Sensors* 2023, 23, 3682. [CrossRef]

- 71. Liang, Y.-C.; Niyato, D.; Larsson, E.G.; Popovski, P. Guest Editorial: 6G Mobile Networks: Emerging Technologies and Applications. *China Commun.* **2020**, *17*, 90–91. [CrossRef]
- 72. Asghar, M.Z.; Memon, S.A.; Hämäläinen, J. Evolution of Wireless Communication to 6G: Potential Applications and Research Directions. *Sustainability* **2022**, *14*, 6356. [CrossRef]
- 73. Polymeni, S.; Plastras, S.; Skoutas, D.N.; Kormentzas, G.; Skianis, C. The impact of 6G-IOT technologies on the development of agriculture 5.0: A Review. *Electronics* 2023, 12, 2651. [CrossRef]
- 74. Liao, Y.; Liu, S.; Hong, X.; Shi, J.; Cheng, L. Integration of communication and navigation technologies toward leo-enabled 6G networks: A survey. *Space Sci. Technol.* **2023**, *3*, 92. [CrossRef]
- 75. Siddiqi, M.A.; Yu, H.; Joung, J. 5G Ultra-Reliable Low-Latency Communication Implementation Challenges and Operational Issues with IoT Devices. *Electronics* **2019**, *8*, 981. [CrossRef]
- 76. The Impact of 5G URLLC Business Connectivity. T-Mobile. Available online: https://www.t-mobile.com/business/resources/ articles/5g-urllc (accessed on 21 February 2024)
- Exploring the Impact of 5G on Telecommunications. Utilities One. Available online: https://utilitiesone.com/exploring-the-impact-of-5g-on-telecommunications (accessed on 23 February 2024).
- 6GWorld. (2024, February 15). Artificial Intelligence for 6G Technology. 6GWorld. Available online: https://www.6gworld.com/ exclusives/artificial-intelligence-for-6g/ (accessed on 23 February 2024).
- Shen, L.-H.; Feng, K.-T.; Hanzo, L. Five facets of 6G: Research challenges and opportunities. ACM Comput. Surv. 2023, 55, 1–39. [CrossRef]
- 80. Benn, H. Hyper Connectivity with 6G. TelecomTV, 2020. Available online: https://www.telecomtv.com/content/what-next-forwireless-infrastructure-summit/hyper-connectivity-with-6g-40147/ (accessed on 23 February 2024).
- 81. Alraih, S.; Shayea, I.; Behjati, M.; Nordin, R.; Abdullah, N.F.; Abu-Samah, A.; Nandi, D. Revolution or Evolution? Technical Requirements and Considerations towards 6G Mobile Communications. *Sensors* **2022**, *22*, 762. [CrossRef]
- 82. How 5G and 6G Technologies Are Transforming Cloud Gaming? Tech Insight. Available online: https://tech.analyticsinsight. net/how-5g-and-6g-technologies-are-transforming-cloud-gaming/ (accessed on 22 February 2024).
- 83. de Alwis, C.; Pham, Q.-V.; Liyanage, M. 6G for Healthcare. In *6G Frontiers: Towards Future Wireless Systems*; IEEE: New York, NY, USA, 2023; pp. 189–196. [CrossRef]
- 84. Wang, C.; He, T.; Zhou, H.; Zhang, Z.; Lee, C. Artificial Intelligence Enhanced Sensors—Enabling Technologies to Next-Generation Healthcare and Biomedical Platform. *Bioelectron. Med.* **2023**, *9*, 17. [CrossRef] [PubMed]
- Aliouche, H. What Is Remote Surgery/Telesurgery? 11 November 2021. Available online: https://www.news-medical.net/ health/What-is-Remote-SurgeryTelesurgery.aspx (accessed on 24 February 2024).
- 86. Kharche, S.; Kharche, J. 6G Intelligent Healthcare Framework: A Review on Role of Technologies, Challenges and Future Directions. *J. Mob. Multimed.* **2023**, *19*, 603–644. [CrossRef]
- 87. Alsharif, M.H.; Jahid, A.; Kannadasan, R.; Kim, M.-K. Unleashing the potential of sixth generation (6G) wireless networks in smart energy grid management: A comprehensive review. *Energy Rep.* **2024**, *11*, 1376–1398. [CrossRef]
- Ahsan, F.; Dana, N.H.; Sarker, S.K.; Li, L.; Muyeen, S.M.; Ali, M.F.; Tasneem, Z.; Hasan, M.d.M.; Abhi, S.H.; Islam, M.R.; et al. Data-driven next-generation smart grid towards Sustainable Energy Evolution: Techniques and Technology Review. *Prot. Control. Mod. Power Syst.* 2023, *8*, 43. [CrossRef]
- 89. Hu, H.; Chen, X.; Jiang, T. Guest Editorial: Brain-Computer-Interface Inspired Communications. *China Commun.* **2022**, *19*, iii–v. [CrossRef]
- 90. Singh, G. Wireless Brain-Computer Interactions (BCI) and 6G Connectivity. Telecom Trainer. 16 March 2023. Available online: https://www.telecomtrainer.com/wireless-brain-computer-interactions-bci/ (accessed on 3 November 2023).
- 91. Pajooh, H.H.; Demidenko, S.; Aslam, S.; Harris, M. Blockchain and 6G-Enabled IoT. Inventions 2022, 7, 109. [CrossRef]
- Hewa, T.; Gür, G.; Kalla, A.; Ylianttila, M.; Bracken, A.; Liyanage, M. The Role of Blockchain in 6G: Challenges, Opportunities and Research Directions. In Proceedings of the 2020 2nd 6G Wireless Summit (6G SUMMIT), Levi, Finland, 17–20 March 2020; pp. 1–5. [CrossRef]
- 93. Dicandia, F.A.; Fonseca, N.J.G.; Bacco, M.; Mugnaini, S.; Genovesi, S. Space-Air-Ground Integrated 6G Wireless Communication Networks: A Review of Antenna Technologies and Application Scenarios. *Sensors* **2022**, *22*, 3136. [CrossRef]
- 94. Alwis, C.D.; Kalla, A.; Pham, Q.-V.; Kumar, P.; Dev, K.; Hwang, W.-J.; Liyanage, M. Survey on 6G Frontiers: Trends, Applications, Requirements, Technologies and Future Research. *IEEE Open J. Commun. Soc.* **2021**, *2*, 836–886. [CrossRef]
- 95. Siddiki Abir, M.d.A.; Chowdhury, M.Z.; Jang, Y.M. Software-Defined UAV Networks for 6G Systems: Requirements, Opportunities, Emerging Techniques, Challenges, and Research Directions. *IEEE Open J. Commun. Soc.* **2023**, *4*, 2487–2547. [CrossRef]
- 96. Hou, Z.; She, C.; Li, Y.; Niyato, D.; Dohler, M.; Vucetic, B. Intelligent Communications for Tactile Internet in 6G: Requirements, Technologies, and Challenges. *IEEE Commun. Mag.* 2021, *59*, 82–88. [CrossRef]
- You, X.; Wang, C.-X.; Huang, J.; Gao, X.; Zhang, Z.; Wang, M.; Huang, Y.; Zhang, C.; Jiang, Y.; Wang, J.; et al. Towards 6G Wireless Communication Networks: Vision, Enabling Technologies, and New Paradigm Shifts. *Sci. China Inf. Sci.* 2020, 64, 110301. [CrossRef]
- Padhi, P.K.; Charrua-Santos, F. 6G Enabled Industrial Internet of Everything: Towards a Theoretical Framework. *Appl. Syst. Innov.* 2021, 4, 11. [CrossRef]

- 99. Qadir, Z.; Le, K.N.; Saeed, N.; Munawar, H.S. Towards 6G Internet of Things: Recent Advances, Use Cases, and Open Challenges. ICT Express 2023, 9, 296–312. [CrossRef]
- Chakrabarti, K. Deep Learning-Based Offloading for Mobile Augmented Reality Application in 6G. Comput. Electr. Eng. 2021, 95, 107381. [CrossRef]
- Admin. The Impact of 6G on Virtual and Augmented Reality. isp.page. 2023. Available online: https://isp.page/news/theimpact-of-6g-on-virtual-and-augmented-reality/ (accessed on 5 November 2023).
- 102. Qiao, L.; Li, Y.; Chen, D.; Serikawa, S.; Guizani, M.; Lv, Z. A Survey on 5G/6G, AI, and Robotics. *Comput. Electr. Eng.* 2021, 95, 107372. [CrossRef]
- 103. Deng, X.; Wang, L.; Gui, J.; Jiang, P.; Chen, X.; Zeng, F.; Wan, S. A Review of 6G Autonomous Intelligent Transportation Systems: Mechanisms, Applications and Challenges. J. Syst. Archit. 2023, 142, 102929. [CrossRef]
- 104. Gallego-Madrid, J.; Sanchez-Iborra, R.; Ortiz, J.; Santa, J. The Role of Vehicular Applications in the Design of Future 6G Infrastructures. *ICT Express* 2023, *9*, 556–570. [CrossRef]
- 105. How 6G Networking Will Solve Your City's Traffic Problems. Available online: https://www.avnet.com/wps/portal/us/ resources/article/how-6g-networking-will-solve-traffic-problems/ (accessed on 4 November 2023).
- Gupta, R.; Reebadiya, D.; Tanwar, S. 6G-enabled Edge Intelligence for ultra-reliable low latency applications: Vision and mission. *Comput. Stand. Interfaces* 2021, 77, 103521. [CrossRef]
- 107. Corici, M.-I.; Eichhorn, F.; Bless, R.; Gundall, M.; Lindenschmitt, D.; Bloessl, B.; Petrova, M.; Wimmer, L.; Kreuch, R.; Magedanz, T.; et al. Organic 6G networks: Vision, requirements, and research approaches. *IEEE Access* **2023**, *11*, 70698–70715. [CrossRef]
- 108. M21. 6G Promises Immersive Communications for Public Safety. Mobility21. 22 May 2023. Available online: https://mobility21. cmu.edu/6g-promises-immersive-communications-for-public-safety/ (accessed on 20 February 2024).
- 109. Tataria, H.; Shafi, M.; Dohler, M.; Sun, S. Six Critical Challenges for 6G Wireless Systems: A Summary and Some Solutions. *IEEE Veh. Technol. Mag.* 2022, 17, 16–26. [CrossRef]
- 110. Zhaona, W.; Kenta, U.; Janne, L.; Nizar, Z. Device-to-device Communications at the Terahertz Band: Open Challenges for Realistic Implementation. *IEEE Commun. Stand. Mag.* 2023, 7, 82–87. [CrossRef]
- 111. Kharche, S.; Dere, P. Interoperability Issues and Challenges in 6G Networks. J. Mob. Multimed. 2022, 18, 1445–1470. [CrossRef]
- Electronics Sourcing. 2023, April 13. 6G Is Happening, and Here's What You Need to Know. Available online: https://electronicssourcing.com/2023/04/13/6g-is-happening-and-heres-what-you-need-to-know/ (accessed on 8 November 2023).
- 113. Sri, A.; Suryanegara, M. Visible Light Communication (VLC) for 6G Technology: The Potency and Research Challenges. In Proceedings of the 2020 Fourth World Conference on Smart Trends in Systems, Security and Sustainability (WorldS4), London, UK, 27–28 July 2020; pp. 490–493. [CrossRef]
- imec. 6G Energy Efficiency. Available online: https://www.imec-int.com/en/articles/boost-6g-energy-efficiency-we-needmodels-can-handle-its-complexities (accessed on 8 November 2023).
- Wang, M.; Zhu, T.; Zhang, T.; Zhang, J.; Yu, S.; Zhou, W. Security and Privacy in 6G Networks: New Areas and New Challenges. Digit. Commun. Netw. 2020, 6, 281–291. [CrossRef]
- 116. Raman, R.; Ravi Kumar, R.; Garg, N.; Joshi, K.; Pillai, B.G.; Joshi, U. Analysis of Potential Health and Environmental Risks Associated with 6G Wireless Communication Networks. In Proceedings of the 3rd International Conference on Advance Computing and Innovative Technologies in Engineering (ICACITE), Greater Noida, India, 12–13 May 2023; pp. 852–856. [CrossRef]
- 117. Nine Takeaways from Early 6G Research-Ericsson. Ericsson. Available online: https://www.ericsson.com/en/blog/2023/2/6g-early-research-global-takeaways (accessed on 23 February 2024).
- 118. Khatib, E.J.; Alvarez-Merino, C.S.; Luo-Chen, H.Q.; Moreno, R.B. Designing a 6G testbed for location: Use cases, challenges, enablers and Requirements. *IEEE Access* 2023, *11*, 10053–10091. [CrossRef]
- 119. Sen, P.; Pados, D.A.; Batalama, S.N.; Einarsson, E.; Bird, J.P.; Jornet, J.M. The Teranova Platform: An Integrated Testbed for Ultra-Broadband Wireless Communications at True Terahertz Frequencies. *Comput. Netw.* **2020**, *179*, 107370. [CrossRef]
- Sun, H.; Ng, C.; Huo, Y.; Hu, R.Q.; Wang, N.; Chen, C.-M.; Vasudevan, K.; Yang, J.; Montlouis, W.; Ayanda, D.; et al. Massive MIMO. In Proceedings of the 2022 IEEE Future Networks World Forum (FNWF), Montreal, QC, Canada, 10–14 October 2022; pp. 1–51. [CrossRef]
- 121. Gera, B.; Raghuvanshi, Y.S.; Rawlley, O.; Gupta, S.; Dua, A.; Sharma, P. Leveraging AI-enabled 6G-driven IoT for sustainable smart cities. *Int. J. Commun. Syst.* 2023, *36*, e5588. [CrossRef]
- 122. Nokia. Spectrum for 6G Explained. Available online: https://www.nokia.com/about-us/newsroom/articles/spectrum-for-6G-explained/ (accessed on 28 November 2023).
- 123. Shahjalal, M.; Kim, W.; Khalid, W.; Moon, S.; Khan, M.; Liu, S.; Lim, S.; Kim, E.; Yun, D.-W.; Lee, J.; et al. Enabling technologies for AI empowered 6G massive radio access networks. *ICT Express* 2023, *9*, 341–355. [CrossRef]
- 124. Alliance. 6G Technologies. Available online: https://www.nextgalliance.org/wp-content/uploads/dlm_uploads/2022/07/ TWG-report-6G-technologies.pdf (accessed on 28 November 2023).
- 125. Fayad, A.; Cinkler, T.; Rak, J.; Jha, M. Design of Cost-Efficient Optical Fronthaul for 5G/6G Networks: An Optimization Perspective. *Sensors* 2022, 22, 9394. [CrossRef] [PubMed]

- 126. Spectrum Options and Allocations for 6G: A Regulatory and Standardization Review. Available online: https://eprints.whiterose. ac.uk/202287/1/Spectrum_Options_and_Allocations_for_6G_A_Regulatory_and_Standardization_Review.pdf (accessed on 28 November 2023).
- 127. Slimani, K.; Khoulji, S.; Kerkeb, M.L. Advancements and challenges in energy-efficient 6G Mobile Communication Network. *E3s Web Conf.* **2023**, *412*, 01036. [CrossRef]
- 128. Taneja, A.; Saluja, N.; Taneja, N.; Alqahtani, A.; Elmagzoub, M.A.; Shaikh, A.; Koundal, D. Power Optimization Model for Energy Sustainability in 6G Wireless Networks. *Sustainability* **2022**, *14*, 7310. [CrossRef]
- 129. The Smart City Journal. 6G: Global Standards vs. Fragmented Ecosystems. The Smart City Journal. 29 December 2021. Available online: https://www.thesmartcityjournal.com/en/articles/6g-global-standards-vs-fragmented-ecosystems (accessed on 27 November 2023)
- Müller, S. Building Trust and Security for 6G Networks with DPI. The Fast Mode. 9 December 2022. Available online: https://www.thefastmode.com/expert-opinion/29386-building-trust-and-security-for-6g-networks-with-dpi (accessed on 21 February 2024).
- Hexa, X. Data Protection Evolution in 6G—Ericsson. Available online: https://www.ericsson.com/en/blog/2023/10/hexa-xand-data-protection-evolution-in-6g (accessed on 28 November 2023).
- Morra, J. Engineers Look to Adopt a More Sustainable Approach to Electronic Design. Electronic Design. 10 November 2023. Available online: https://www.electronicdesign.com/resources/industry-insights/article/21275020/electronic-designengineers-look-to-adopt-a-more-sustainable-approach-to-electronic-design (accessed on 7 November 2023).
- 133. Basic Information about Electronics Stewardship | US EPA. Available online: https://www.epa.gov/smm-electronics/basicinformation-about-electronics-stewardship (accessed on 29 November 2023).
- 134. Edge Computing and Embedded Artificial Intelligence. ECS SRIA. Available online: https://ecssria.eu/2.1 (accessed on 23 February 2023).
- 135. Dong, X.J. An Innovative Technology for 6G Communication Networks. Tech Xplore—Technology and Engineering News. 2022. Available online: https://techxplore.com/news/2022-02-technology-6g-networks.html (accessed on 17 November 2023).
- 136. Huang, Y.; Shen, Y.; Wang, J. From Terahertz Imaging to Terahertz Wireless Communications. *Engineering* **2023**, *22*, 106–124. [CrossRef]
- 137. Jiang, W.; Zhang, Q.; He, J.; Habibi, M.A.; Melnyk, S.; El-Absi, M.; Han, B.; Renzo, M.D.; Schotten, H.D.; Luo, F.-L.; et al. Terahertz Communications and Sensing for 6G and Beyond: A Comprehensive View. *TechRxiv* 2023, *1*, 8–10. [CrossRef]
- 138. Aslam, T.; Ahmed, I.; Ali, S.; Aslam, M.I. Terahertz Communication and Associated Challenges in 6G Cellular Networks. In Proceedings of the 4th International Conference on Computing and Information Sciences (ICCIS), Phagwara, India, 4–5 December 2021; pp. 1–6. [CrossRef]
- Hall, R. 5G and Gan: Understanding Sub-6ghz Massive Mimo Infrastructure. Embedded.com. 26 January 2021. Available online: https://www.embedded.com/5g-and-gan-understanding-sub-6ghz-massive-mimo-infrastructure/ (accessed on 24 November 2023).
- 140. Chowdhury, M.Z.; Shahjalal, M.; Ahmed, S.; Jang, Y.M. 6G Wireless Communication Systems: Applications, Requirements, Technologies, Challenges, and Research Directions. *IEEE Open J. Commun. Soc.* **2020**, *1*, 957–975. [CrossRef]
- 141. Ning, B.; Tian, Z.; Mei, W.; Chen, Z.; Han, C.; Li, S.; Yuan, J.; Zhang, R. Beamforming Technologies for Ultra-Massive MIMO in Terahertz Communications. *IEEE Open J. Commun. Soc.* **2023**, *4*, 614–658. [CrossRef]
- 142. Pérez Santacruz, J.; Meyer, E.; Budé, R.X.F.; Stan, C.; Jurado-Navas, A.; Johannsen, U.; Tafur Monroy, I.; Rommel, S. Outdoor MM-wave 5G/6G Transmission with Adaptive Analog Beamforming and IFOF Fronthaul. Nature News. 25 August 2023. Available online: https://www.nature.com/articles/s41598-023-40112-w (accessed on 25 November 2023).
- 143. Kassam, J.; Castanheira, D.; Silva, A.; Dinis, R.; Gameiro, A. A Review on Cell-Free Massive MIMO Systems. *Electronics* **2023**, 12, 1001. [CrossRef]
- 144. He, H.; Yu, X.; Zhang, J.; Song, S.; Letaief, K.B. Cell-Free Massive MIMO for 6G Wireless Communication Networks. *J. Commun. Inf. Netw.* **2021**, *6*, 321–335. [CrossRef]
- 145. He, S.; Du, J.; Liao, Y. Multi-User Scheduling for 6G V2X Ultra-Massive MIMO System. Sensors 2021, 21, 6742. [CrossRef]
- 146. Ajmal, M.; Siddiqa, A.; Jeong, B.; Seo, J.; Kim, D. Cell-Free Massive Multiple-Input Multiple-Output Challenges and Opportunities: A Survey. *ICT Express* **2023**, *10*, 194–212. [CrossRef]
- 147. Dala Pegorara Souto, V.; Dester, P.S.; Soares Pereira Facina, M.; Gomes Silva, D.; de Figueiredo, F.A.P.; Rodrigues de Lima Tejerina, G.; Silveira Santos Filho, J.C.; Silveira Ferreira, J.; Mendes, L.L.; Souza, R.D.; et al. Emerging MIMO Technologies for 6G Networks. Sensors 2023, 23, 1921. [CrossRef]
- 148. Moltchanov, D.; Sopin, E.; Begishev, V.; Samuylov, A.; Koucheryavy, Y.; Samouylov, K. A Tutorial on Mathematical Modeling of 5G/6G Millimeter Wave and Terahertz Cellular Systems. *IEEE Commun. Surv. Tutor.* **2022**, *24*, 1072–1116. [CrossRef]
- Ethw. Millimeter Waves. ETHW. 12 April 2017. Available online: https://ethw.org/Millimeter_Waves (accessed on 23 October 2023).
- 150. Qu, K.; Chen, K.; Zhao, J.; Zhang, N.; Hu, Q.; Zhao, J.; Jiang, T.; Feng, Y. An Electromechanically Reconfigurable Intelligent Surface for Enhancing Sub-6G Wireless Communication Signal. *J. Inf. Intell.* **2023**, *1*, 207–216. [CrossRef]
- 151. Sharma, T.; Chehri, A.; Fortier, P. Reconfigurable Intelligent Surfaces for 5G and beyond Wireless Communications: A Comprehensive Survey. *Energies* 2021, *14*, 8219. [CrossRef]
- 152. Henrique, P.S.; Prasad, R. 6G Networks Orientation by Quantum Mechanics. J. ICT Stand. 2022, 10, 39-62. [CrossRef]

- 153. Nawaz, S.J.; Sharma, S.K.; Wyne, S.; Patwary, M.N.; Asaduzzaman, M. Quantum Machine Learning for 6G Communication Networks: State-of-the-Art and Vision for the Future. *IEEE Access* **2019**, *7*, 46317–46350. [CrossRef]
- 154. Shrestha, R.; Bajracharya, R.; Kim, S. 6G enabled Unmanned Aerial Vehicle Traffic Management: A perspective. *IEEE Access* 2021, 9, 91119–91136. [CrossRef]
- 155. Kukliński, S.; Szczypiorski, K.; Chemouil, P. UAV Support for Mission Critical Services. Energies 2022, 15, 5681. [CrossRef]
- 156. Telli, K.; Kraa, O.; Himeur, Y.; Ouamane, A.; Boumehraz, M.; Atalla, S.; Mansoor, W. A Comprehensive Review of Recent Research Trends on Unmanned Aerial Vehicles (UAVs). *Systems* **2023**, *11*, 400. [CrossRef]
- 157. Farhad, A.; Pyun, J.-Y. Terahertz Meets AI: The State of the Art. Sensors 2023, 23, 5034. [CrossRef]
- 158. Khan, M.M.; Hossain, S.; Mozumdar, P.; Akter, S.; Ashique, R.H. A review on machine learning and deep learning for various antenna design applications. *Heliyon* 2022, *8*, e09317. [CrossRef]
- 159. Alhaj, N.A.; Jamlos, M.F.; Manap, S.A.; Abdelsalam, S.; Bakhit, A.A.; Mamat, R.; Jamlos, M.A.; Gismalla, M.S.M.; Hamdan, M. Integration of Hybrid Networks, AI, Ultra Massive-MIMO, THz Frequency, and FBMC Modulation Toward 6G Requirements: A Review. *IEEE Access* 2024, 12, 483–513. [CrossRef]
- Huo, Y.; Lin, X.; Di, B.; Zhang, H.; Hernando, F.J.L.; Tan, A.S.; Mumtaz, S.; Demir, Ö.T.; Chen-Hu, K. Technology Trends for Massive MIMO towards 6G. Sensors 2023, 23, 6062. [CrossRef] [PubMed]
- 161. Rekkas, V.P.; Sotiroudis, S.; Sarigiannidis, P.; Wan, S.; Karagiannidis, G.K.; Goudos, S.K. Machine Learning in Beyond 5G/6G Networks—State-of-the-Art and Future Trends. *Electronics* **2021**, *10*, 2786. [CrossRef]
- Lu, Y.; Jiang, H.; Dai, L. Artificial Intelligence for RIS-Aided Wireless Communications. ITU J. Future Evol. Technol. 2023, 4, 70–77. [CrossRef]
- 163. Ali, M.Z.; Abohmra, A.; Usman, M.; Zahid, A.; Heidari, H.; Imran, M.A.; Abbasi, Q.H. Quantum for 6G Communication: A Perspective. *IET Quantum Commun.* **2023**, *4*, 112–124. [CrossRef]
- 164. Yin, S.; Zhao, S.; Zhao, Y.; Yu, F.R. Intelligent Trajectory Design in UAV-Aided Communications with Reinforcement Learning. *IEEE Trans. Veh. Technol.* **2019**, *68*, 8227–8231. [CrossRef]
- 165. Webb, M.D. Risks to Human Health: High-Frequency Radio Waves. 30 August 2019. Available online: https://www.myfanwywebb.com/5g-6g-risks-to-human-health-high-frequency-radio-waves/ (accessed on 25 November 2023).
- US EPA. Electric and Magnetic Fields from Power Lines. Available online: https://www.epa.gov/radtown/electric-and-magnetic-fields-power-lines (accessed on 25 November 2023).
- ITU. 5G, Human Exposure to Electromagnetic Fields (EMF) and Health. Available online: https://www.itu.int/en/mediacentre/backgrounders/Pages/5G-EMF-health.aspx (accessed on 25 November 2023).
- Simkó, M.; Mattsson, M.-O. 5G Wireless Communication and Health Effects—A Pragmatic Review Based on Available Studies Regarding 6 to 100 GHz. Int. J. Environ. Res. Public Health 2019, 16, 3406. [CrossRef]
- 169. Center for Devices and Radiological Health. Cell Phones. U.S. Food and Drug Administration. Available online: https://www.fda.gov/ radiation-emitting-products/home-business-and-entertainment-products/cell-phones (accessed on 25 November 2023).
- Patriciello, N.; Lagen, S.; Giupponi, L.; Bojovic, B. 5G New Radio Numerologies and their Impact on the End-To-End Latency. In Proceedings of the 2018 IEEE 23rd International Workshop on Computer Aided Modeling and Design of Communication Links and Networks (CAMAD), Barcelona, Spain, 17–19 September 2018; pp. 1–6. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.